Lawrence Berkeley National Laboratory

LBL Publications

Title

Remote Sensing-Based Forest Modeling Reveals Positive Effects of Functional Diversity on Productivity at Local Spatial Scale

Permalink

https://escholarship.org/uc/item/0kp437b6

Journal

Journal of Geophysical Research Biogeosciences, 128(6)

ISSN

2169-8953

Authors

Schneider, Fabian D Longo, Marcos Paul-Limoges, Eugénie <u>et al.</u>

Publication Date

2023-06-01

DOI

10.1029/2023jg007421

Copyright Information

This work is made available under the terms of a Creative Commons Attribution-NonCommercial License, available at <u>https://creativecommons.org/licenses/by-nc/4.0/</u>

Peer reviewed

Remote sensing-based forest modeling reveals positive effects of functional diversity on productivity at local spatial scale

Fabian D Schneider^{1,2}, Marcos Longo^{3,1}, Eugénie Paul-Limoges², Victoria M Scholl^{4,5,2}, Bernhard Schmid², Felix Morsdorf², Ryan P Pavlick¹, David S Schimel¹, Michael E Schaepman²& Paul R Moorcroft⁶

¹Jet Propulsion Laboratory, California Institute of Technology, Pasadena, CA 91109, United States ²Remote Sensing Laboratories, Department of Geography, University of Zurich, CH-8057 Zurich, Switzerland

³Lawrence Berkeley National Laboratory, Climate and Ecosystem Sciences Division, Berkeley, CA, 94720, United States

⁴Earth Lab, Cooperative Institute for Research in Environmental Sciences, University of Colorado

Boulder, Boulder, CO 80303, United States ⁵Department of Geography, University of Colorado Boulder, Boulder, CO 80309, United States ⁶Department of Organismic and Evolutionary Biology, Harvard University, Cambridge, MA 02138, United

States

Key Points:

- We explored forest functional diversity-productivity relationships using a terrestrial biosphere model initialized with remote sensing
- Higher functional diversity at 20×20 -m to 30×30 -m scale was associated with higher productivity (p < 0.001) across all simulations
- Soil depth and texture were the main drivers of productivity at the hectare scale, while functional diversity saturated at coarser scales

Corresponding author: Fabian D. Schneider, fabian.schneider@jpl.nasa.gov

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1029/2023JG007421.

Abstract

Forest biodiversity is critical for many ecosystem functions and services. Yet, it remains uncertain how plant functional diversity influences ecosystem functioning across environmental gradients and contiguous larger areas. We integrated remote sensing and terrestrial biosphere modeling to explore functional diversity-productivity relationships at multiple spatial scales for a heterogeneous forest ecosystem in Switzerland. We initialized forest structure and composition in the ecosystem demography model (ED2) through a combination of ground-based surveys, airborne laser scanning and imaging spectroscopy for forest patches at 10×10 -m spatial grain. We derived morphological and physiological forest traits and productivity from model simulations at patch-level to relate morphological and physiological aspects of functional diversity to the average productivity from 2006-2015 at 20×20 -m to 100×100 -m spatial extent. We did this for model simulations under observed and experimental conditions (mono-soils, mono-cultures and monostructures). Functional diversity increased productivity significantly (p < 0.001) across all simulations at 20×20 -m to 30×30 -m scale, but at 100×100 -m scale positive relationships disappeared under homogeneous soil conditions potentially due to the low beta diversity of this forest and the saturation of functional richness represented in the model. Although local functional diversity was an important driver of productivity, environmental context underpinned the variation of productivity (and functional diversity) at larger spatial scales. In this study, we could show that the integration of remotely-sensed information on forest composition and structure into terrestrial biosphere models is important to fill knowledge gaps about how plant biodiversity affects carbon cycling and biosphere feedbacks onto climate over large contiguous areas.

Plain Language Summary

We explored relationships between forest biodiversity and productivity at multiple spatial scales to better understand how plant traits and their diversity influence the functioning of forest ecosystems. To do this across large contiguous areas, we simulated a temperate mixed forest by combining process-based terrestrial biosphere modeling with remotely sensed information about forest structure and composition. We found that higher functional diversity was associated with higher productivity at local spatial scale (tens of meters), while soil depth and texture were the main drivers of productivity at the hectare scale. At this larger scale, functional diversity tended to be saturated due to the low turnover of species and functional traits of the temperate mixed forest and the limited representation of physiological diversity in the model.

1 Introduction

Biodiversity is a key property of forests that affects important ecosystem services, including provisioning services such as timber supply or water purification, cultural services of recreational and spiritual value, and regulating services such as carbon sequestration (Hooper et al., 2005; Cardinale et al., 2012; Chamagne et al., 2017; Aerts et al., 2018; Isbell et al., 2017). Biodiversity also plays an important role in the carbon cycle and influences vegetation-atmosphere interactions and feedbacks (Schimel et al., 2015). For example, higher biodiversity may increase the resistance and resilience of ecosystems to climate change and increase carbon sequestration contributing to climate change mitigation (Isbell et al., 2015; Liang et al., 2016; J. Liu et al., 2018; Huang et al., 2018). However, earth's biodiversity is declining rapidly with negative consequences for ecosystem functions and services (Cardinale et al., 2012; Ceballos et al., 2015; Díaz et al., 2019a). Thus, losses in biodiversity may contribute to accelerating impacts of climate change on forest ecosystems (Mori, 2020), triggering potentially additional species losses (Trisos et al., 2020; Arneth et al., 2020). Despite recognized importance of plant diversity to help ecosystems provide multiple functions and services (Hautier et al., 2018; Díaz et al., 2019b), 21698951, j, Downloaded from https://agupubs.onlineibary.wiley.com/doi/10.10292023/G07421 by Univ or California Lawrence Berkeley National Lab, Wiley Online Library on [0905/2023]. See the Terms and Conditions (https://alinelibary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Comm

there is still considerable uncertainty in understanding and predicting the mechanistic relationship between biodiversity and ecosystem functioning, such as productivity or biomass production, (BEF) and the role of different aspects of biodiversity such as functional, genetic, phylogenetic or taxonomic diversity. Specifically, there is a need to improve our understanding of how BEF relationships scale up from plot-level to large contiguous areas of forest, how BEF relationships may be modified by environmental variation and change, which measures of biodiversity can be used for large-scale assessments, and how biodiversity effects are represented in mechanistic demographic modeling of forest dynamics that can be used for predicting future scenarios and changes.

Positive effects of species richness on ecosystem functioning and stability have been reported widely at plot scale, while mostly focusing at biomass production as an indicator of ecosystem functioning and stability. Initially, results were published mainly from experimental studies (Balvanera et al., 2006; Cardinale et al., 2011; Tilman et al., 2014) but recently also from observational studies (e.g. Grace et al., 2016; Liang et al., 2016; Duffy et al., 2017; van der Plas, 2019). Jochum et al. (2020) showed that results from grassland experiments represent realistic species compositions and Liang et al. (2016); Ochri et al. (2020) found positive effects when controlling for differences in climate across sites, but it is unclear whether the relationships will hold within natural ecosystems that extend beyond the typical plot-scale (Gonzalez et al., 2020), and very few studies explicitly test plot-size effects (e.g. Huang et al., 2018; Chisholm et al., 2013; Poorter et al., 2015). Therefore, scaling up BEF relationships remains a major challenge. We expect BEF relationships to span multiple scales, but the form and strength of the relationship might change across scales, biomes and diversity metrics (Gonzalez et al., 2020). There have been a number of studies addressing environmental context in BEF studies (see Hong et al., 2022, for a recent meta-analysis), but few focused on forests. Paquette and Messier (2011), Ratcliffe et al. (2017), and Mina et al. (2017) found that the mechanisms and BEF effects varied with environmental conditions and stress. Yet, whether or not forests will remain a carbon sink under future climate remains uncertain (Brienen et al., 2015; Sabatini et al., 2019; Arora et al., 2020; Hubau et al., 2020), and this uncertainty may be at least partially attributable to BEF.

Previous BEF literature explored which components of biodiversity are most predictive for ecosystem functioning, but no single component is consistently the best (Craven et al., 2018). Functional diversity is often considered promising (Loreau, 2000; Hooper et al., 2005), and in a review of 258 BEF studies of naturally assembled communities, van der Plas (2019) found that, while most studies focused on the effects of taxonomic diversity, metrics of functional diversity were generally stronger predictors of ecosystem functioning. This is in line with early findings from grassland experiments. For example, Tilman, Knops, et al. (1997) found that functional composition and functional diversity were the principal factors explaining plant productivity and Petchey et al. (2004) found that functional diversity explained greater variation in biomass production than species richness. Yet, forest functional diversity is rarely assessed explicitly and independent of taxonomy (but see Schneider et al., 2017; Guillén-Escribà et al., 2021). Most plant studies thus far have focused on the effects of species richness (as a measure of biodiversity) on biomass (as a measure of ecosystem function) in grassland ecosystems, mostly under uniform abiotic conditions. Therefore, little is known about other dimensions of diversity and function, and their relationships in forests across a larger range of environmental gradients and spatial scales (Sheil & Bongers, 2020).

Regarding underlying mechanisms behind BEF relationships, the theoretical reasoning suggests that niche complementarity (complementary traits lead to better resource use of a community as a whole), selection probability (higher likelihood to have highly productive individuals), and ecological insurance (higher likelihood to adapt to or recover from fluctuating environmental conditions or disturbance) underpin biodiversity effects on ecosystem functioning and stability (Tilman, Lehman, & Thomson, 1997; Yachi & Loreau, 1999; Loreau, 2000). However, integration of observed biodiversity patterns into terrestrial biosphere models and direct demographic modeling of these mechanisms still represents a challenge, but it is critically important.

Simulations and model experiments using process-based models can help fill some of these knowledge gaps, especially since diversity experiments and measurements of plant functional traits, diversity and functioning are difficult to conduct in forest ecosystems. Morin et al. (2011), for example, used a process-based dynamic vegetation model to simulate biodiversity experiments for a range of sites and species in Europe, finding consistent positive effects of biodiversity on productivity due to complementarity. Extending this approach to larger scales, Levine et al. (2016), Sakschewski et al. (2016) and Longo et al. (2018) found that ecosystem heterogeneity and diversity were important to predict forest resilience to climate change including extreme drought events in the Amazon. Many terrestrial biosphere models, however, are not able to resolve this level of detail, or only allow to simulate potential vegetation, which can lead to uncertainties in the prediction of the carbon cycle (Braghiere et al., 2019) and climate change feedbacks (Schimel et al., 2019; Fisher & Koven, 2020). The integration of remote observations of forest composition and structure, such as from imaging spectroscopy and lidar, into models holds great promise to improve our understanding of patterns and drivers of forest diversity and productivity. Previous studies have shown that incorporating such measurements into biosphere models, such as the ecosystem demography model version 2 (ED2), substantially improved their predictions of forest carbon fluxes (A. Antonarakis, 2014; A. S. Antonarakis et al., 2022; Rödig et al., 2018).

In this study, we investigate the relationship between forest functional diversity, as a measure of biodiversity, and productivity, as a measure of ecosystem functioning, across spatial scales and environmental gradients by integrating remotely sensed measures of forest structure and composition in the terrestrial biosphere model ED2 (Fig. 1). Analyses of the model outputs enabled us to address the two main research questions: (1) What is the relationship between functional diversity and productivity at different spatial scales? (2) How does the relationship between diversity and productivity vary across environmental and ecological gradients involving different soil types, plant functional types (PFTs) and canopy structural types? We expect a positive relationship between functional diversity and productivity over longer time scales, since diversity promotes spatial and temporal complementarity and stability in light and water use as simulated among different plant functional and structural types in ED2 (Longo et al., 2020; Meunier et al., 2021; A. S. Antonarakis et al., 2022). We also expect differences in soil types to influence both diversity and productivity, especially if water availability is limited. For example, simulations by Longo et al. (2018) showed that forests on sandier soils were more resilient to drought, and Medvigy et al. (2019) found that variation in soil properties caused variation in biomass and ecosystem composition in simulations of forest succession. Based on Chisholm et al. (2013), we would expect sampling effects and niche complementarity to dominate at small scales, while environmental gradients should drive patterns at large scales. However, this has not been tested in terrestrial biosphere models and in observational studies it is often difficult to disentangle effects of composition and environment.

Materials and methods

We combine field data with high-spatial resolution airborne laser scanning and imaging spectroscopy measurements to derive estimates of temperate forest structure (i.e. stem density and diameter at breast height) and composition (i.e. plant functional types) that we use as initial conditions in the process-based terrestrial biosphere model ED2 (Fig. 1, Longo, Knox, Medvigy, et al., 2019). ED2 models dynamic interactions, niche complementarity and competition for resources (light and water) (Longo, Knox, Medvigy, et al., 2019; Longo, Knox, Levine, et al., 2019). We simulate gross primary productiv-



21698961, ja, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.10292023JG007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Library on [09/05/2023]. See the Terms and Conditions (https://online

elibrary. wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Comm



Figure 1. Forest modeling approach to derive forest diversity and productivity relationships in a temperate mixed forest. We used airborne imaging spectroscopy and light detection and ranging (LiDAR) together with field inventory data to derive stem density, tree diameters at breast height (DBH) and plant functional types (PFT). These variables were used to represent the forest in the ED2 terrestrial biosphere model in 10×10 m patches and vertically resolved cohorts within patches, on different soil types. Forest productivity was then simulated as gross primary productivity (GPP) based on hourly meteorological drivers and averaged over 10 years. Together with traits retrieved from the model (morphological: basal area (BA), leaf area index (LAI), foliage height diversity (FHD), and physiological: VC_{max}, specific leaf area (SLA)), this allowed us to relate the morphological and physiological diversity to the simulated long-term forest productivity. The visualization shows an example based on data and simulations conducted in this study.

This article is protected by copyright. All rights reserved.

ity (GPP) under observed meteorological boundary conditions from 2006 to 2015 (Fig. 1) for forests across a range of observed soil depths and textures (Fig. 2). We evaluate the model's ability to simulate carbon dynamics and seasonality of the studied temperate mixed forest as measured at the eddy covariance flux tower at the Laegern forest in Switzerland. To explore what is driving the spatial variation in forest GPP at 20×20 to 100×100 -m spatial scale, we run and analyze simulations based on observed soil, forest structure and composition as well as model experiments where we test combinations of homogenized soil types (mono-soils), plant functional types (mono-cultures), and canopy structural types (mono-structures). We then use the model simulation outputs to investigate the role of different aspects of forest functional diversity on spatial patterns of forest productivity at local to landscape scales (Fig. 1). For this, we analyzed the simulated diversity-productivity relationships using general linear models and random forest models. In terms of functional diversity metrics, we investigate the role of richness, divergence and evenness (see Schneider et al., 2017) derived from morphological traits, namely leaf area index (LAI), basal area (BA) and foliage height diversity (FHD), and physiological traits, namely maximum leaf Rubisco carboxylation rate (VC_{max}) and specific leaf area (SLA), modeled at cohort-level. In the following sections, we provide a detailed description and methodology of the study area, the terrestrial biosphere model, the integration of remote sensing data into the model, model evaluation and simulations, and how we derived and analyzed the model outputs with regard to morphological and physiological diversity and productivity.

Study area

We conducted this study at the Laegern temperate mixed forest in Switzerland (47°28'43.0 N, 8°21'53.2 E), which is located on a mountain range representing broad ecological and environmental gradients (Fig. 2). The study area is characterized by increasing soil depth (40 to 75 cm) and soil grain size (clayey to sandy loam), and mean tree stem diameter and abundance of conifers from high to low elevation (850–450 m above sea level). The vegetation consists of diverse beech forests with a total of 13 tree species and about 70% deciduous broadleaf and 30% evergreen coniferous tree individuals. The dominant deciduous species are common beech (*Fagus sylvatica* L.), European ash (*Fraxinus excelsior* L.), and sycamore maple (*Acer pseudoplatanus* L.). The dominant coniferous species are Norway spruce (*Picea abies* (L.) H. Karst) and silver fir (*Abies alba* Mill.).

Laegern is a key ecological and remote sensing site in central Europe (Morsdorf et al., 2020), which has been studied extensively in terms of 3D structure and radiative transfer (Schneider et al., 2014; Fawcett et al., 2018; Schneider et al., 2019; Kükenbrink et al., 2021), functional and genetic diversity (Schneider et al., 2017; Czyz et al., 2020; Guillén-Escribà et al., 2021; Helfenstein et al., 2022; Czyż et al., 2023), and carbon and water fluxes (Paul-Limoges et al., 2018; Damm et al., 2020; Paul-Limoges et al., 2020). The study area has a fluxtower (CH-LAE; Paul-Limoges et al., 2020) equipped with eddy-covariance instruments and a meteorological station, as well as a 5.5-ha field plot, where 1307 canopy trees were mapped by stem location, crown extent, diameter at breast height (DBH) and species identity (Morsdorf et al., 2020; Guillén-Escribà et al., 2021).

We selected five regions to study diversity-productivity relationships, covering the range of soil characteristics and the ecological gradient from shorter deciduous broadleaf forest on top of the ridge to taller mixed communities in lower elevations. Each region was sub-divided into three 100×100 -m sites, selected to minimize environmental heterogeneity within regions and maximize the range of abiotic and biotic variation between regions and sites, respectively (Fig. 2).

-6-



Figure 2. Study regions at the Laegern mountain, Switzerland, covering a gradient in soil, structure, composition and diversity of the temperate mixed forest. The three-dimensional surface topography is colored by the four plant functional types: late-successional conifers, early-, mid- and late-successional hardwoods. Regions 1–5 are each comprised of three 1-ha sites (white boxes) and show a shift from shorter deciduous broadleaf forest on top of the mountain (1) to taller evergreen coniferous forest at lower elevation (5) following a gradient in soils from shallow clayey loam (1) to deeper sandy loam (5). The histograms show the distribution of basal area per tree diameter at breast height (DBH) for the four plant functional types. The Laegern fluxtower is located at the north-east corner of region three (pink triangle). Model evaluation was performed using forest composition and structure representative of the fluxtower footprint, shown as black outline at 70% flux density. The map (top-right) shows the location of Laegern north-west of Zurich, with respect to the country border of Switzerland.

This article is protected by copyright. All rights reserved.

Model description

The ED2 model is a process-based terrestrial biosphere model that represents individual plant-level dynamics (growth, mortality and recruitment), and associated ecosystemlevel carbon, water and energy fluxes over time scales ranging from hours to centuries (Fig.1; Longo, Knox, Medvigy, et al., 2019). In contrast to conventional 'ecosystem as big leaf' models that represent the canopy in a highly aggregated manner, the ecosystem demography (ED) model utilizes the concept introduced by Moorcroft et al. (2001), in which vegetation is represented as cohorts of plants of similar height and plant functional type, grouped in patches that can be initialized with spatially explicit remote observations, within a site of homogeneous environmental conditions with regard to soil, topography and meteorology (Longo, Knox, Medvigy, et al., 2019). Forest patches and cohorts are initialized by PFTs, which have constant plant functional traits associated to them that do not vary within PFT, and DBH and stem density, which vary dynamically over time. Plant growth, mortality and recruitment dynamics are simulated as part of vegetation dynamics. The model can be run with vegetation dynamics off, where only phenological and physiological dynamics are simulated.

The size- and age-structured representation allows us to incorporate fine-scale spatial variation in canopy structure and composition arising from disturbance processes and spatially-localized, size-dependent competition for light within the above-ground plant canopy and below-ground for water (bigger trees have deeper roots and thus can access water in deeper soil layers). To improve vegetation dynamics regarding canopy structure and light availability compared to previous ED modeling schemes, heterogeneity in horizontal and vertical micro-climate environments has been introduced in ED2 (Medvigy et al., 2009; Longo, Knox, Medvigy, et al., 2019). The ED2 biosphere model has shown good conservation of energy, water and carbon (Longo, Knox, Medvigy, et al., 2019; Longo, Knox, Levine, et al., 2019) and has been successfully run with initial conditions from foreststand data as observed from remote sensing (Longo et al., 2020), field plots (Meunier et al., 2021) or a combination of both (Bogan et al., 2019; Meunier et al., 2022).

Integration of remote sensing data into the model

The ED2 model can be initialized by prescribing the plant functional types (PFTs), tree stem diameter (DBH) and stem density organized as tree cohorts within forest patches. We derived forest composition (PFTs) and structure (DBH and stem density) using a combination of field data, airborne imaging spectroscopy, and laser scanning (ALS) at high spatial resolution of 2×2 -m (Fig. S1). We then integrated these empirical data into the model as initial conditions of tree cohorts within forest patches of 10×10 -m (Figs. S2, S3).

Forest composition was integrated into the model based on a classification of ED2-PFTs originally defined for Harvard forest (Medvigy et al., 2009), namely: late-successional conifers and early-, mid-, and late-successional hardwoods (LCf, EHw, MHw, LHw, respectively). These four ED2-PFTs and corresponding physiological traits have been extensively used and tested in temperate forests in North America and Europe (e.g. Medvigy et al., 2009, 2010; A. Antonarakis, 2014; Jin et al., 2017; Paul-Limoges et al., 2020). A. Antonarakis (2014) noted the importance of high-resolution PFT classification and forest structure estimates to capture fine-scale ($<10\times10$ -m) PFT-specific variation in forest canopy composition and structure. In this study, we address this requirement by performing the PFT classification at high (2×2 -m) spatial resolution using a random forest classifier with remotely sensed input features, trained and validated based on field data of 1307 identified trees (Morsdorf et al., 2017). From 81 possible input features, we selected 27 relevant features (see Table 1) that yielded the best model results, including physiological and morphological forest traits (leaf chlorophylls, carotenoids, water and **Table 1.** Remote sensing input features used for the random forest classification of plant functional types. We used three morphological and three physiological forest traits from Schneider et al. (2017), and their filtered version applying a median filter with a window size of 7×7 pixels. We also used selected principal components of surface reflectance and continuum-removed reflectance from APEX, data described in Supporting Note 1 and Helfenstein et al. (2022).

| Random forest feature | Remote Sensing | |
|--|-------------------------|--|
| Canopy height (CH) | Airborne Laser Scanning | |
| CH 7×7 median filtered | Airborne Laser Scanning | |
| Plant Area Index (PAI) | Airborne Laser Scanning | |
| PAI 7×7 median filtered | Airborne Laser Scanning | |
| Foliage Height Diversity (FHD) | Airborne Laser Scanning | |
| FHD 7×7 median filtered | Airborne Laser Scanning | |
| Leaf Chlorophylls (CHL) | Imaging Spectroscopy | |
| CHL 7×7 median filtered | Imaging Spectroscopy | |
| Leaf Carotenoids (CAR) | Imaging Spectroscopy | |
| CAR 7×7 median filtered | Imaging Spectroscopy | |
| Leaf Water (LWC) | Imaging Spectroscopy | |
| LWC 7×7 median filtered | Imaging Spectroscopy | |
| Reflectance principal components 1-9, 13, 18 | Imaging Spectroscopy | |
| Continuum-removed reflectance principal components 1, 2, 4, 10 | Imaging Spectroscopy | |

canopy height, plant area index, foliage height diversity, see Schneider et al., 2017), principal components of surface reflectance and continuum-removed reflectance acquired by the airborne imaging spectrometer APEX (Schaepman et al., 2015). The imaging spectroscopy data was processed following Hueni et al. (2009, 2013) and geometrically corrected following Schläpfer and Richter (2002), for more details see Supporting Note 1. The classifier had an overall accuracy of 74% and a Cohen's kappa coefficient of 61% for the classification of plant functional types over the whole forest. Further details are provided on data in Supporting Note 1, Schneider et al. (2017); Helfenstein et al. (2022), on the classification approach in Supporting Note 2, on the remote sensing workflow in Fig. S1, and on the random forest classification accuracy in Fig. S4. The classified map of plant functional types is shown in Fig. S5.

Forest structure was integrated into the model as initial conditions of DBH and stem density estimated from ALS and field data. DBH values were estimated from ALSderived canopy height using an exponential model that was fitted to field inventory data of 159 late conifers, 253 early hardwoods, 328 mid hardwoods and 566 late hardwoods:

$$DBH = a \cdot CH^{b} + \epsilon(\sigma, \gamma, \kappa), \tag{1}$$

where CH is canopy height, a and b are coefficients and ϵ is the residual distribution around the mean. To predict not only the mean but a realistic DBH distribution, we derived the standard deviation σ , skewness γ and kurtosis κ of the distribution to model ϵ (Matlab R2017a, pearsrnd). The resulting values for a, b, σ, γ and κ are summarized in Table S1. We applied this approach to predict a DBH value for each 2×2-m pixel based on its

This article is protected by copyright. All rights reserved.

remotely sensed canopy height. Single-tree delineation was applied to the ALS point cloud to estimate the density of stems of different sizes (i.e. canopy height classes) across each 1-ha site (Morsdorf et al., 2004; Kaartinen et al., 2012; Y. Wang et al., 2016). The method had a detection rate of 79% overall, 83% for trees ≥ 35 m and 76% for trees < 35 m. We show a comparison between the field data and predicted stem density and DBH in Figures S6 and S7. We then assigned stem density to each 2×2-m pixel based on canopy height, by assigning the average density estimate per 1-m canopy height class derived at each study region. For more information on the forest structure estimation, see Supporting Note 1 and Kükenbrink et al. (2017) for the underlying lidar data, see Supporting Note 3 for more details on the stem segmentation and basal area calculation, and see Figure S1 for a workflow overview.

Finally, values were aggregated as cohorts in discrete 10×10 -m forest patches across each 100×100 -m site by averaging DBH and stem density for each plant functional type and 10-cm DBH class (Fig. S2), which were used as initial conditions in the model (Fig. S3). This approach reduces the number of cohorts to simulate in the model, reduces computational expenses, and better approximates the individual tree scale of the forest while preserving plant functional type distributions within patches. The cohort-based nature of the ED2 model means that forest canopy structure can be prescribed from spatiallyresolved field or remote sensing measurements of DBH and stem density. Therefore, the current state of the forest's morphological diversity can be directly incorporated into the model simulation, thereby accounting for impacts of past disturbances and forest management on canopy structure. All other structural attributes, such as tree height, leaf area index (LAI) and biomass, are then calculated by ED2 using the model's allometric relationships that are part of the definition of each PFT (Fig. S3).

Meteorological drivers should reflect short-term (hourly-to-weekly) meteorological variability, which has an important effect on ecosystem functioning such as carbon sequestration, and, over decadal timescales, can drive shifts in forest structure and composition (Medvigy et al., 2010). Therefore, we used hourly meteorological variables measured between 2006 and 2015 as boundary conditions. Atmospheric pressure at field elevation (hPa), air temperature (°C), relative humidity (%), average wind speed (m s⁻¹), and precipitation (mm) were measured at 10-minute intervals on the Laegern fluxtower as part of the national air pollution monitoring network (NABEL) of Switzerland. Incoming shortwave (W m⁻²) and longwave radiation fluxes (W m⁻²) were measured at 1-minute intervals. For gap filling of missing values (<2%), we used meteorological data from nearby meteorological stations (Supporting Note 4, Fig. S8). All meteorological variables were averaged to one hour for input into the ED2 model. Atmospheric CO₂ was assumed constant at 391 µmol mol⁻¹, consistent with the average background values observed at Mauna Loa between 2006 and 2015 (Tans & Keeling, 2020).

Soil information was used from FOAG (1996) to specify spatial variation in soil depth and texture (relative sand, loam, clay content) as the main ED2 inputs (Fig. 2).

Model evaluation

We evaluated the model's carbon flux predictions for the period 2006–2015 against corresponding estimates from eddy-covariance fluxes (Baldocchi, 2003) measured continuously at the Laegern flux tower. Flux quality post-processing was done following Vickers and Mahrt (1997). Standardized gap filling and partitioning of the net ecosystem CO_2 exchange into GPP and ecosystem respiration were done using the method from Barr et al. (2004). We ran the flux footprint prediction model FFP (Kljun et al., 2015) to estimate the average proportions of PFTs, DBH and stem density within the footprint based on the 2×2-m remote sensing maps (Fig. S1). The predicted average daytime fluxtower footprint is shown in Figure S9. We then selected a sample of 200 representative forest Table 2. Model simulations under observed initial conditions and experiments run using ED2 on five regions each comprised by three 1-ha sites (15 total) to produce monthly outputs for the years 2006–2015. Multiple entries separated by | represent multiple simulations in a set. Mono-soils were simulated at three depths (40, 51, 75 cm) and three textures (.-clayey loam, o-loam, O-sandy loam), and the combinations thereof. Monocultures were simulated for early-, mid- and late-successional hardwoods (EHw, MHw, LHw) and late-successional conifers (LCf). Mono-structures were simulated for sparse, medium and dense canopies.

| Simulation | Composition | Structure | Soil | Vegetation dynamics |
|------------------------------|------------------------|--------------------------|--|------------------------|
| Observed | as observed | as observed | as observed | on $ $ off |
| Experiment Mono-soil | as observed | as observed | 40 . 40 o 40 O 51 . 51 o 51 O 75 . 75 o 75 O | off |
| Experiment Mono-culture | EHw MHw LHw LCf | as observed | 40 . 51 o 75 O | off |
| Experiment Mono-structure | as observed | sparse medium dense | 40 . 51 o 75 O | off |

patches for simulation in ED2 and comparison to fluxtower estimates that resembled the average forest composition within the footprint as shown in Figure S10.

We ran a 10-year spin-up phase using the 200 patches as initial condition before simulating the years 2006–2015 including vegetation dynamics. Longo, Knox, Levine, et al. (2019) tested the sensitivity of the model with regard to different spin up times from 5 to 60 years and found that the effect was relatively small if the spin up time was at least 5 years. Demographic rates showed low sensitivity. Also, GPP and the dynamic of the living trees is not influenced by soil carbon pools, which would take longer to reach equilibrium. Other studies used similar spin up times. For example, A. S. Antonarakis and Guizar Coutino (2017) used a 10 year period and Medvigy et al. (2009) used a spin up of 3 years for grasses.

We also assessed the predicted phenology by comparing ED2's monthly simulated LAI with satellite-based LAI estimated using the data assimilation model PhenoAnalysis (Stöckli et al., 2011). We ran the PhenoAnalysis model in the data assimilation mode to fit an LAI time series to the noisy Moderate Resolution Imaging Spectroradiometer (MODIS) satellite data taking into account the gap filled meteorological variables from the fluxtower (Fig. S11).

Model simulations and experiments

For all simulations, we ran the model over a 20-year period by cycling through the meteorological drivers time series twice. The first 10-year cycle was considered spin-up, so that the soil moisture and the vegetation dynamics could stabilize after integrating initial conditions of the forest, and only the second 10-year cycle was considered in the analyses.

To address the two research questions and disentangle the effects of forest structure, composition and soil on productivity, we ran five sets of simulations on the 15 (5×3) 100×100 -m sites, summarized in Table 2. The first set of simulations were run with the observed composition, structure, and soil information for the 15 sites, with both vegetation dynamics on and off. Note that with vegetation dynamics off, phenological and physiological dynamics are still simulated, but the dynamics of mortality, growth and recruitment are only simulated when vegetation dynamics are on. We then ran three experimental simulations to which we refer as mono-soils, mono-cultures and mono-structures. In the mono-soil simulations, the 15 sites were run with observed stand composition and structures on nine uniform soils (3 soil depths \times 3 soil textures). In the mono-culture simulations, the 15 sites were run with four different uniform PFTs, each on 3 soils (i.e. 4 PFTs \times 3 soils \times 15 sites). Finally, the mono-structure simulations were run with the observed composition across the 15 sites, but using three different uniform canopy structures, each on 3 soils (3 structures \times 3 soils \times 15 sites). Canopy structure types were created by replicating the vegetation structure of forest patches that had low (19 m² ha⁻¹), medium (39 m² ha⁻¹) and high (83 m² ha⁻¹) basal area across all 15 sites, while keeping the PFT-composition as observed. The model experiments were run with vegetation dynamics off to maintain the specified canopy composition, structure and soil type combinations, which might not naturally occur.

21698961, ja, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023JG007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Library on [0905/2023]. See the Terms and Conditions (https://online

elibrary.wiley.com/terms-

and-conditions) on Wiley Online Library for rules of use; OA articles

s are governed by the applicable Creative Comm

Forest functional traits, diversity and productivity

Forest functional traits were calculated as patch-averages at 10×10 -m spatial grain from cohort-level ED2 model outputs. Traits and diversity remained constant in simulations with vegetation dynamics off, but can vary when vegetation dynamics are simulated. To improve the comparability and interpretation of the various results, we related productivity to initial functional diversity of summer 2006. This marks the first peak growing season of the time series and the initial condition as prescribed by remote sensing and field data, thus providing valuable insights into how initial diversity predicts productivity over time (Hagan et al., 2021). The selected traits represent important ecological and model variables for ecosystem functioning, relating to the distribution of leaf area (LAI), tree diameter and density (basal area, BA), vertical canopy layering (foliage height diversity, FHD), maximum leaf Rubisco carboxylation rate (VC_{max}) and specific leaf area (SLA). The physiological traits VC_{max} and SLA were derived from the model values assigned to each PFT and within-patch PFT-mixtures by calculating weighted averages based on the cohort leaf area index and leaf biomass, respectively (Fig. S3). This means that the resulting values of VC_{max} and SLA per forest patch were restricted to the mixtures of four PFTs and their trait values, which are constant in the model. The morphological traits were also derived based on the cohorts within each forest patch, with the difference that they can vary within and between PFTs (based on remotely sensed tree diameter and density, and PFT-specific model allometries to LAI and height) and dynamically over time (if vegetation dynamics is simulated). BA and LAI were derived as cohort averages per patch, and FHD was calculated from the vertical distribution of cohort LAI values within each patch following Schneider et al. (2017). We then calculated multi-dimensional functional diversity metrics from those traits (see Fig. S3), specifically morphological and physiological richness, divergence and evenness at 20×20 , 30×30 , 40×40 , 50×50 and 100×100 -m spatial extent following Schneider et al. (2017). We defined functional richness as the average of morphological and physiological richness, since both should receive equal weight but they were calculated with a different number of traits. For forest productivity, we used cohort-level outputs to calculate monthly GPP per 10×10^{-10} m forest patch and average annual and decadal GPP of 2006–2015. We also calculated GPP normalized by leaf area, for comparison of leaf-level productivity.

This article is protected by copyright. All rights reserved.

Statistical analyses

To study the effects of morphological and physiological diversity on GPP, we fitted a linear model (Matlab 2019a, fitlm) with functional diversity metrics (log(richness), divergence, evenness) as explanatory variables and log(GPP) as the response variable; and a general linear model (Matlab 2019a, fitglme) with soil depth*texture combinations and functional diversity metrics as fixed effects. We also analyzed the model outputs with a random forest model to show the relative feature importance of average functional traits and diversity for predicting GPP at 20×20 and 100×100 -m spatial resolution. The analysis with and without including average plant functional traits in the random forest model indicates contributions of functional diversity for predicting GPP in a direct (additional to mean traits) and indirect (due to correlations to mean traits) way. Predictor importance was estimated from permutations of out-of-bag predictor observations of 1000 regression trees (Matlab 2019a, fitrensemble, oobPermutedPredictorImportance; Schneider et al., 2020), and scaled relative to the fraction of variance explained (r^2) by the model.

We analyzed the effects of morphological and physiological richness on productivity in model experiments (mono-soil, mono-culture, mono-structure). We fitted general linear models (Matlab 2019a, fitglme) with log(GPP) as response variable, and soil depth*texture combinations and log(richness) as fixed effects in mono-soil, soil depth*texture, PFT and log(richness) in monoculture, and soil depth*texture, structure and log(richness) in monostructure experiments, respectively.

Results

Integrating high-resolution remote observations of forest composition and structure into the terrestrial biosphere model ED2 allowed us to predict the field-measured size-structure as basal area per diameter class with an r^2 of 0.94, ranging from 0.70 to 0.93 for individual plant functional types at the flux tower plot. Figure S12 shows that we were able to capture the distribution of basal area per DBH and PFT in the model, and Figure S13 shows the scatter plots and linear model results compared to the fieldbased estimates. This allowed us to study the relationships between morphological and physiological diversity and forest productivity at various spatial scales (20–100-m) along an environmental gradient (5 soil types) in simulations of observed conditions and model experiments. We first evaluated the model performance by comparing simulated carbon fluxes with eddy covariance-based estimates at the temperate mixed forest's fluxtower over the course of ten years, which built the basis for the subsequent diversity–productivity analyses.

Model evaluation

We found that the model was able to accurately predict the carbon dynamics and seasonality of the Laegern temperate mixed forest averaged across the flux tower footprint, explaining 86% and 82% of monthly and diurnal variations in GPP as estimated at the flux tower using eddy-covariance data from 2006 to 2015. Figure 3 shows the simulated carbon dynamics with respect to the most important meteorological drivers and flux tower estimates. The model captured the seasonality of monthly GPP, but was sensitive to water available to the plants, potentially overestimating their physiological response.

For example, annual GPP of the year 2011 was underestimated by 27% (Fig. 3a) following an exceptionally dry spring with precipitation 63% below the decadal average (Fig. 3b). The increased water stress was also reflected by the model parameter FSW (ecosystem-scale down-regulation factor for photosynthesis due to limited soil water availability; FSW = 1 means no soil water stress), which showed values 13% below the growing season average, indicating increased drought stress. Similar effects of reduced GPP



Figure 3. ED2 model evaluation of monthly and diurnal carbon fluxes and monthly leaf area index (LAI) against eddy covariance tower-based and satellite-based estimates in the context of meteorological drivers. The panels show (a) monthly simulated gross primary productivity (GPP, solid red line), heterotrophic and plant respiration (light green area), net ecosystem productivity (NEP, dark green area, dotted red line), the flux tower estimates of GPP (solid gray line) and NEP (dotted gray line); (b) monthly radiation (red area), precipitation (blue area), percent days with snow cover (white bars), and air temperature (solid gray line); (c) modeled GPP (ED2) against flux tower estimates (Tower) per month and (d) as monthly average diurnal cycle per hour; (e) modeled LAI (ED2) as monthly average of site-level LAI against satellite-based LAI using PhenoAnalysis to retrieve LAI from Moderate Resolution Imaging Spectroradiometer (MODIS). The colorscale of wetter to drier months is based on a z-score transformation of monthly precipitation values from 2006–2015 to a mean of 0 and a standard deviation of 1 for each month.

-14-

and FSW were observed for dry summer months in 2006, 2008 and 2014, and an early GPP drop-off in dry and hot summers of 2013 and 2015. In 2006, the wet spring and fall (115% and 24% above average precipitation) led to 23% higher annual GPP than estimated by the tower despite hot and dry June and July. In summary, the model overestimated tower GPP under wetter conditions and underestimated it under drier conditions, at both monthly (Fig. 3c) and diurnal scales (Fig. 3d), but generally captured the dynamics with little overall bias. Seasonality was generally well predicted, with a slight tendency to an extended productivity at the end of season. This is also reflected in a higher LAI in October in most years due to ED2 predicting delayed leaf fall in comparison with satellite-based LAI (Fig. 3e).

Here, we focus on GPP, but the model shows a good overall performance in simulating carbon fluxes including net ecosystem productivity as estimated by the flux tower $(r^2 = 0.76)$, even though respiration is difficult to estimate from fluxes at this particular site due to the steep slope, dense upper canopy and potential below canopy CO₂ drainage (Paul-Limoges et al., 2017; Etzold et al., 2010).

Diversity-productivity across sites

We found generally positive relationships between functional diversity and productivity across sites, spatial scales, and functional diversity indices when modeling GPP with vegetation dynamics and observed soil types. A general linear model showed highly significant effects of local morphological and physiological richness (p < 0.001) and soil types (p < 0.001) on average GPP at 20×20-m to 50×50-m scale, when controlling for different soils between regions (Fig. 4a,b,d,e, Fig. S14; Tab. S2, S3). There is a strong separation in GPP between regions 1-2 and the rest, following differences in soil matric potential and depth (Fig. 4i). Functional divergence and evenness effects were less strong and more variable from positive to neutral (Figs. S15, S16, Tabs. S4, S5, S6, S7).

At 100×100-m scale, we found significant positive relationships with productivity across sites for morphological (p < 0.001) and physiological (p < 0.05) richness (Fig. 4c,f). Figure 4g and h show the increase of functional richness and narrowing of the range with spatial extent. Morphological richness was significantly different between regions with the shallowest and deepest soils (Fig. 4g), while physiological richness did not differ significantly except between region 1 and 3 at 100×100-m (Fig. 4h).

A random forest model analysis showed that environmental variables related to soil water were the dominant predictors of GPP at 100×100 -m scale, but functional diversity still showed a relative importance of 9–18% and up to 17–20% for predicting GPP when considering indirect effects through correlation to other functional traits (Fig. 5). At local scale, community-weighted average physiological traits had the highest relative importance (51%) when predicting GPP with vegetation dynamics enabled, whereas communityweighted average morphological traits were most important (27%) when vegetation dynamics was disabled (Fig. 5). In both cases, the importance of functional diversity strongly increased when both direct and indirect effects were considered (11–55% with vegetation dynamics on, 9–26% with vegetation dynamics off). Functional richness was the overall most important diversity metric, and both morphological and physiological richness were important and varied significantly with GPP (Figs. 4, 5).

Diversity-productivity model experiments

In natural ecosystems, co-varying drivers of diversity and productivity, such as different soil characteristics, make it difficult to draw conclusions about diversity effects on productivity (Loreau, 2000; Grace et al., 2016). The model environment enabled us to run experiments with homogeneous soils (mono-soils), plant functional types (mono-cultures)



Figure 4. Relationship between (a-c) morphological and (d-f) physiological richness and productivity as average annual gross primary productivity (GPP), simulated across five regions over the course of a decade with vegetation dynamics at (a,d) 20x20-m, (b,e) 50-50-m, and (c,f) 100x100-m scale. Statistics in (a,b,d,e) are based on a general linear model controlling for different soils between regions, while (c,f) shows a linear regression across all sites. Panels (g,h) show the overlap of richness for each region across spatial scales. Significant differences between the regions within three spatial scales are indicated by the compact letter display based on Fisher's least significant difference test. Panel (i) shows soil matric potential as indicator of soil water available to the plants (more negative numbers mean more strength required to extract water), and corresponding soil depth of the five regions. Colors are consistent across panels and refer to the five different regions and their unique soil properties.



Figure 5. Random forest feature importance to predict average annual gross primary productivity, based on simulation outputs with vegetation dynamics on (a,b) and off (c,d) at 20 m (a,c) and 100 m (b,d) spatial scale, scaled by the total variation explained (r^2) by the model. The gray bars indicate unexplained variance. The white numbers indicate the relative importance of a group of variables, while the partitioning within the group has to be interpreted with care due to possibly stronger correlations among the different explanatory variables in a group than between the groups. By including (top of each spatial scale) and excluding (bottom of each spatial scale) average functional traits, we can see direct and indirect effects of functional diversity on productivity.

and canopy structural types (mono-structures) to disentangle the effects of soil, morphological and physiological richness on long-term productivity.

We found that functional richness, as average of morphological and physiological richness, on mono-soils had a significant positive effect on GPP (p < 0.001) at local scale (20–30 m, see Fig. 6a, Tab. S8). Only when considering productivity per leaf area, the richness effect remained strongly positive among all scales (p < 0.001 at 20–100 m, Fig. S17, Tab. S9). Evenness and divergence showed mixed results from positive to negative and highly to non-significant relationships without clear patterns (Fig. S18).

The mono-culture simulations further showed that the effect of morphological richness on productivity was highly significant at local scales (p < 0.001 at 20–40 m, Tab. S10), but the effect varied by PFT, with no effect for late-successional hardwoods and the most positive effect for late-successional conifers (Figs. 6b, Fig. S19). A positive effect was also found for physiological richness across canopy structures (Fig. 6c; Tab. S11), with strongest positive effect for tall and dense canopies at scales from 20–50 m (Fig. S20).

In all three experiments, the maximum difference in productivity between the lowest and highest productivity soils, plant functional types and canopy structures, respectively, was larger than between lowest and highest forest functional diversity (Fig. 6). Also, there was no diversity effect at $100 \times$ -m scale in the experiments (Figs. S17, S19, S20), indicating that morphological and physiological diversity were especially important at local scale.



Figure 6. Boxplots comparing low, medium and high (a) functional (FRic), (b) morphological (MRic) and (c) physiological richness (PRic) with (a) shallow, medium and deep soil, (b) late-successional conifers (LCf), late-, early- and mid-successional hardwoods (LHw, EHw, MHw) plant functional types (PFT), and (c) sparse, medium and dense canopy structural types (CST), respectively, at 20×20 -m spatial extent. Diversity values were split into low, medium and high based on percentiles at 1/3 and 2/3 of the distributions. Simulations were run with single soil type (mono-soil, each 375 areas at 20×20 m), single PFTs (monoculture, 375×3 soils), and CSTs (mono-structure, 375×3 soils). The scatter plots show all data points and linear regression lines for (a) three soil textures (clayey loam, loam, sandy loam) per soil depth, (b) three soils per PFT and (c) three soils per CST. The three soils were clayey loam at 40 cm, loam at 51 cm, and sandy loam at 75 cm depth. P-values are shown based on general linear models controlling for soil.

Discussion

Our modeling results show that there are significant positive effects of morphological and physiological diversity on average annual forest productivity in the 2006–2015 period at local spatial scales (tens of meters), and that this phenomenon holds both within soil types and across an environmental gradient. Among diversity indices, we found the strongest and most consistent relationship between functional richness and productivity, while evenness and divergence showed weaker and less consistent results. Functional richness is a measure of the total niche space covered in the multivariate trait space, which can indicate resource use efficiency and utilization through the range of functional strategies and growth forms (Cadotte et al., 2011; Schneider et al., 2017). Therefore, we expect it to be a good predictor of productivity (Durán et al., 2019). Since evenness and divergence describe the distribution within the occupied niche, they are less informative in small niches or with small sample sizes but they can contribute to explaining variation in forest productivity if richness is large (Fig. 5). These positive diversity effects are likely stronger in reality, because terrestrial biosphere models do not include all processes and interactions between species and forest patches, such as the plastic responses of species traits to environmental conditions and stand diversity (Roscher et al., 2018).

At larger spatial scales $(100 \times 100 \text{-m})$, we found overall positive correlations between functional diversity and productivity along the environmental gradient of differing soil

textures and depth, comparable to Durán et al. (2019). This effect is more complex, however, because soils may act as confounding factors as evidenced by the loss of the hectarescale diversity-productivity relationship when simulating uniform soil types. In the following sections, we discuss the processes and mechanisms in the terrestrial biosphere model that drive the influence of morphological, physiological and environmental aspects of the forest ecosystem on its functioning.

Morphological diversity and productivity

Morphological diversity of the plant canopy structure controls the radiative transfer and light distribution throughout the canopy, with higher diversity in canopy structure leading to a more even distribution of light throughout the canopy and higher light use efficiency and productivity (Ishii et al., 2004; Williams et al., 2017; Kükenbrink et al., 2021). In the ED2 model, higher morphological diversity also relates to higher complementarity in soil water use, as plant size (i.e. tree diameter and height) determines the plant rooting depth. High local variability in plant size means high variability in rooting depth, thus reducing competition for water in the same soil layer, increasing complementarity and overall productivity through reduced stress (Loreau et al., 2001).

The effect of morphological diversity on productivity is likely underestimated in the model, since interactions between forest patches, such as changed radiative transfer through shading or sun exposure from neighboring patches, were not modeled. However, the remotely sensed structural information that we used as initial conditions in the model likely did integrate processes that are not modeled explicitly, such as nutrient cycles, wind throw disturbance or forest management. Morphological diversity effects in the model are therefore mostly stemming from observed relationships between diversity and other stand-level structural properties that influence model productivity, such as increased stem density and basal area in structurally more diverse forest areas as observed in other studies (Barrufol et al., 2013; Morton et al., 2016). Such a limitation is common in terrestrial biosphere models (Fisher & Koven, 2020) and shows the importance of integrating empirical observations of forest structural variability into the models (e.g., Caylor et al., 2004; Hurtt et al., 2010; Fischer et al., 2016; Levine et al., 2016; Longo et al., 2018; Braghiere et al., 2019). Additional efforts are needed to map forest structure and composition repeatedly and to operationalize data-model integration and model benchmarking over time.

Finally, while tree diameters and density can vary freely and they can be integrated into the model as initial conditions, other structural variables such as height or biomass are defined by static PFT-specific allometric relationships. ALS measurements show, however, a high degree of variability in diameter-height relationships. Incorporating plastic allometry is an interesting area for future model development, since tree heights (and many other variables of canopy structure) can be derived from laser scanning at different spatial scales (e.g. Schneider et al., 2014, 2019, 2020).

Physiological diversity and productivity

The diversity of physiological traits (here only between PFTs, within PFTs these traits are assumed to be constant) represents the variety of functional strategies and life forms (Reich et al., 2003; Díaz et al., 2015), influencing an ecosystem's ability for adaptation, defense and recovery that will determine its long-term stability and productivity (Díaz & Cabido, 2001; Mason & de Bello, 2013). There are two main effects causing productivity to increase as physiological diversity increases that are captured in the model: (1) a temporal stability and insurance effect, and (2) a selection and dominance effect driven by the physiological properties of the model PFTs.

With respect to (1), the temporal asynchrony of plant communities, species or functional types has been shown to positively influence ecosystem productivity and stability (Tilman et al., 2014; Loreau & de Mazancourt, 2013; Craven et al., 2018), and this effect is also represented by ED2. For example, late-successional conifers can photosynthesize earlier and later in the growing season during leaf development and senescence of deciduous species, and their productivity is more stable due to lower stomatal conductance that leads to lower drought sensitivity (Fig. 7a). In contrast, deciduous hardwoods are more productive during peak greenness, but react more strongly to meteorological variability (Fig. 7a). The ED2 model's plant functional types represent the different seasonal dynamics and stress responses that can explain positive diversity-effects at decadal temporal scales, consistent with previous ecological modeling experiments (Cardinale et al., 2004).



Figure 7. Annual variability in gross primary productivity (GPP) of plant functional type (PFT) monocultures and mixtures, and contributions of PFT-mixing to physiological richness (PRic) and GPP at different spatial scales. a) Annual relative anomaly compared to the decadal mean GPP of late-successional conifers (LCf), late-, early- and mid-successional hard-woods (LHw, EHw, MHw), simulated in monocultures, compared to simulations using observed PFT-mixtures (Mix) on a 51-cm loamy soil. The blue bars and connecting lines show annual precipitation anomalies for context. b) Cumulative contributions of mixtures of 1-4 PFTs to GPP by physiological richness across spatial scales from 20×20 -m to 100×100 -m extent. PRic shows saturation towards larger scales, where almost all PFTs are present at all sites, even though in different abundances.

With regard to (2), there can be a selection and dominance effect in which dominance by species or plant functional types with particular traits affects ecosystem functioning (selection effect; Loreau et al., 2001). As in most central European temperate forests, in our study area beech is the dominant late-successional hardwood species. We modeled late-successional hardwoods with lower VC_{max} and higher SLA to represent their lower maximum productivity and higher shade tolerance compared to early- or mid-successional hardwoods (Peters, 1992; Heiri et al., 2009, see also Fig. 6). The consequence is that earlyand mid-successional hardwoods have higher productivity if enough light is available but cannot maintain high leaf area and productivity in dense canopies or understory, while late-successional hardwoods can maintain dense, tall canopies and productivity under lower light conditions, but with lower maximum productivity. Therefore, lower-diversity beech forests will have lower productivity than mixed forests with local disturbance and location-specific growing conditions that allow early and mid-successional species to compete, co-exist and be highly productive. This diversity of plant community and structure was captured by the integration of remote observations into the model that capture different development stages and the effect of local disturbance processes, which is characteristic of productive and sustainable forests over time (Cardinale et al., 2004; Silva Pedro et al., 2016; Dolezal et al., 2020).

Overall, physiological diversity effects may also be underestimated in the model, since physiological complementarity in nutrient use, complementarity in soil microbial communities and related co-benefits of microbial diversity, or complementarity in insect defence strategies were not included in the model formulation. In addition, physiological traits were defined by the plant functional type of each tree species, which limits functional diversity in physiological traits to mixtures of typically three to four PFTs, even when the total species richness may be considerably higher (13 tree species at our study site). Opportunities exist to implement more flexible traits, for example in terms of withincanopy trait plasticity with light availability (Berzaghi et al., 2020; Xu et al., 2021). Further development is needed, for example towards flexible traits such as leaf mass per area, nitrogen or V_{cmax} , as proposed by Sakschewski et al. (2015) and adapted to European forests by Thonicke et al. (2020). The combined ability to represent plastic trait variation and integrate it from remote sensing could greatly advance the representation of forest functional diversity (e.g. Schneider et al., 2017; Durán et al., 2019; Z. Wang et al., 2020) and the modeling of carbon fluxes and responses to climate change (Schimel et al., 2019).

Spatial scale and environmental drivers

Our results showed that the diversity effects discussed above primarily act at the local (sub-hectare) spatial scale, where functional diversity varies within a large range from low to medium diversity (Fig. 4g,h). This is related to the community structure of the temperate mixed forest, which has high alpha- and low beta-diversity (Schneider et al., 2017), a situation that is typical for most temperate forests (Swenson, 2012). Additionally, most plant interactions in forests happen at the scale of tens of meters (X. Wang et al., 2015), and thus our results are consistent with findings from field-based studies (Chisholm et al., 2013; Poorter et al., 2015; X. Liu et al., 2018). At larger spatial scales, diversity can be saturated when beta-diversity is low. This can lead to functional redundancy (Rosenfeld, 2002) and a weakening of the diversity–productivity relationship (Jochum et al., 2020). In our study, a saturation of physiological diversity with increasing spatial extent might have occurred earlier than in reality due to the limited number of plant functional types (Fig. 7b) and negligence of intraspecific physiological diversity present at this site (Guillén-Escribà et al., 2021).

The positive correlation between diversity and productivity at the hectare-scale was considerably weakened when the sites were simulated with uniform environmental conditions, indicating co-variation of soil properties with plant diversity and productivity. This suggests that the environmental gradient, characterized by variations in soil depth and texture in the model, was the main driver of productivity at this larger scale, limiting plant productivity through limiting resources such as water availability, while functional diversity was saturated. The effect of soil texture and depth on forest productivity may be overestimated because the model shows a higher-than-observed sensitivity of productivity to plant water availability, which is indicated by monthly and diurnal comparisons to the fluxtower (Fig. 3) and annual GPP anomalies following precipitation (Fig. 7a). The steep, fractured limestone bedrock might offer the trees opportunities to access water below the average soil depth, and this currently cannot be represented by ED2. Longo et al. (2018) also noted that ED2 might overestimate the compactness of clay-rich soils, which could exaggerate water limitation in clayey soils. Therefore, the actual trees at Laegern may exhibit more resistance to water stress than our analysis implies (Paul-Limoges et al., 2020).

Conclusions and outlook

Functional diversity had a significant positive effect on decadal forest productivity at local spatial scales (tens of meters) in terrestrial biosphere simulations in a temperate mixed forest. This result was found across all soil types and including morphological and physiological diversity, with strongest effects for functional richness. This implies that local functional richness, which can be measured with remote sensing across large spatial extents and in remote ecosystems, is an important driver of productivity in temperate mixed forests. This finding is in line with studies that analyzed species richnessproductivity relationships in field plots (Chisholm et al., 2013; Poorter et al., 2015).

At larger spatial scales (hectare-scale and above), our model experiments showed that diversity's influence on productivity saturated and that soil depth and texture appeared to be the main drivers of spatial variation in forest productivity at this scale. We note that the influence of soil properties may be overestimated due to the challenge of modeling soil water availability to the plants in this complex terrain. The diversity saturation may in part be due to the community structure of this temperate forest and the limited representation of functional diversity in the model, i.e. physiological diversity being limited to mixtures of plant functional types. We also note that the saturation of biodiversity– ecosystem functioning relationship at the larger scale does not imply that this relationship is not important in driving global forest productivity. That is, if diversity is low at the smaller scale, this also leads to lower productivity at the large scale.

The integration of spatially explicit remotely-sensed trait information in Earth system models is crucial to study the role of biodiversity in carbon cycling and to predict impacts and feedbacks of climate change. The advancement of spaceborne remote sensing will help to characterize plant functional traits and diversity globally, for example morphological diversity using lidar (Schneider et al., 2020) and radar (Bae et al., 2019) or physiological diversity using imaging spectroscopy (Cawse-Nicholson et al., 2021), and will provide new measurements of ecosystem functioning (Schimel et al., 2019) to initialize and benchmark global terrestrial biosphere models that represent structurally and functionally diverse ecosystems (Fisher et al., 2018).

2 Open Research

The datasets that support the findings of this study are available online at figshare (Schneider et al., 2023): https://doi.org/10.6084/m9.figshare.22705024.v1. The most up-to-date ED2 source code is available at https://github.com/EDmodel/ED2, and the version used for this study is permanently deposited on https://dx.doi.org/10.5281/zenodo.3365659.

Acknowledgments

The research carried out at the Jet Propulsion Laboratory, California Institute of Technology, was under a contract with the National Aeronautics and Space Administration (80NM0018D0004). Government sponsorship is acknowledged. This study was supported by the University of Zurich Research Priority Program on 'Global Change and Biodiversity' (URPP GCB). Support was also provided by the NASA HyspIRI Preparatory Activity NNH11ZDA001N-HYSPIRI grant "Linking Terrestrial Biosphere Models with Remote Sensing Measurements of Ecosystem Composition, Structure, and Function" to PRM. ML was supported by the NASA Postdoctoral Program, administered by Universities Space Research Association under contract with NASA. Additional funding was provided by a Swiss Government Excellence Scholarship awarded to VS through the U.S. Fulbright Student Program. Meteorological data was provided by the Federal Office for the Environment (FOEN), the Swiss Federal Laboratories for Materials Science and Technology (EMPA) and MeteoSchweiz (Climap). Stand polygon data was provided by Aargauisches Geografisches Informationssystem (AGIS), Departement Bau, Verkehr und Umwelt, Abteilung Wald (last updated on 27 February 2015) and by Geographisches Informationssystem (GIS-ZH), Amt für Landschaft und Natur, Abteilung Wald (last updated on 16 September 2015). Soil data corresponds to Bodenkarte Baden (Landeskarte der Schweiz 1:25'000, Blatt 1070), provided by Eidgenössische Forschungsanstalt für Agrarökologie und Landbau (FAL).

References

- Aerts, R., Honnay, O., & Van Nieuwenhuyse, A. (2018). Biodiversity and human health: Mechanisms and evidence of the positive health effects of diversity in nature and green spaces. British Medical Bulletin, 127(1), 5–22. doi: 10.1093/bmb/ldy021
- Antonarakis, A. (2014, 11). Uncertainty in initial forest structure and composition when predicting carbon dynamics in a temperate forest. *Ecological Modelling*, 291, 134-141. Retrieved from http://linkinghub.elsevier.com/retrieve/ pii/S0304380014003780http://dx.doi.org/10.1016/j.ecolmodel.2014.07 .030https://linkinghub.elsevier.com/retrieve/pii/S0304380014003780 doi: 10.1016/j.ecolmodel.2014.07.030
- Antonarakis, A. S., Bogan, S. A., Goulden, M. L., & Moorcroft, P. R. (2022, 3). Impacts of the 2012–2015 Californian drought on carbon, water and energy fluxes in the Californian Sierras: Results from an imaging spectrometry-constrained terrestrial biosphere model. *Global Change Biology*, 28(5), 1823–1852. Retrieved from https://onlinelibrary.wiley.com/doi/10.1111/gcb.15995 doi: 10.1111/gcb.15995
- Antonarakis, A. S., & Guizar Coutino, A. (2017, 11). Regional Carbon Predictions in a Temperate Forest Using Satellite Lidar. *IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing*, 10(11), 4954–4960. Retrieved from http://ieeexplore.ieee.org/document/8013090/ doi: 10.1109/JSTARS.2017.2732058
- Arneth, A., Shin, Y.-j., Leadley, P., Rondinini, C., Bukvareva, E., Kolb, M., ... Saito, O. (2020, 12). Post-2020 biodiversity targets need to embrace climate change. Proceedings of the National Academy of Sciences, 117(49), 30882–30891. Retrieved from http://www.pnas.org/lookup/doi/10.1073/ pnas.2009584117https://pnas.org/doi/full/10.1073/pnas.2009584117 doi: 10.1073/pnas.2009584117
- Arora, V. K., Katavouta, A., Williams, R. G., Jones, C. D., Brovkin, V., Friedlingstein, P., ... Ziehn, T. (2020, 8). Carbon-concentration and carbon-climate feedbacks in CMIP6 models and their comparison to CMIP5 models. *Biogeosciences*, 17(16), 4173-4222. Retrieved from https://bg.copernicus.org/ articles/17/4173/2020/ doi: 10.5194/bg-17-4173-2020
- Bae, S., Levick, S. R., Heidrich, L., Magdon, P., Leutner, B. F., Wöllauer, S., ... Müller, J. (2019, 12). Radar vision in the mapping of forest biodiversity from space. Nature Communications, 10(1), 4757. Retrieved from http://www.nature.com/articles/s41467-019-12737-x doi: 10.1038/s41467-019-12737-x
- Baldocchi, D. D. (2003). Assessing the eddy covariance technique for evaluating carbon dioxide exchange rates of ecosystems: past, present and future. *Global Change Biology*, 9(4), 479–492. Retrieved from http://dx.doi.org/10.1046/ j.1365-2486.2003.00629.x doi: 10.1046/j.1365-2486.2003.00629.x
- Balvanera, P., Pfisterer, A. B., Buchmann, N., He, J.-S., Nakashizuka, T., Raffaelli,D., & Schmid, B. (2006, 10). Quantifying the evidence for biodiversity effects

on ecosystem functioning and services. *Ecology Letters*, 9(10), 1146–1156. Retrieved from http://doi.wiley.com/10.1111/j.1461-0248.2006.00963.x doi: 10.1111/j.1461-0248.2006.00963.x

- Barr, A. G., Black, T. A., Hogg, E. H., Kljun, N., Morgenstern, K., & Nesic, Z. (2004). Inter-annual variability in the leaf area index of a boreal aspen-hazelnut forest in relation to net ecosystem production. Agricultural and Forest Meteorology, 126(3), 237–255. Retrieved from http:// www.sciencedirect.com/science/article/pii/S0168192304001686 doi: https://doi.org/10.1016/j.agrformet.2004.06.011
- Barrufol, M., Schmid, B., Bruelheide, H., Chi, X., Hector, A., Ma, K., ... Niklaus, P. A. (2013, 11). Biodiversity Promotes Tree Growth during Succession in Subtropical Forest. *PLoS ONE*, 8(11), e81246. Retrieved from http:// dx.plos.org/10.1371/journal.pone.0081246https://dx.plos.org/ 10.1371/journal.pone.0081246 doi: 10.1371/journal.pone.0081246
- Berzaghi, F., Wright, I. J., Kramer, K., Oddou-Muratorio, S., Bohn, F. J., Reyer, C. P., ... Hartig, F. (2020, 3). Towards a New Generation of Trait-Flexible Vegetation Models. Trends in Ecology & Evolution, 35(3), 191-205. Retrieved from https://www.sciencedirect.com/science/article/pii/ S0169534719303404https://linkinghub.elsevier.com/retrieve/pii/ S0169534719303404 doi: 10.1016/j.tree.2019.11.006
- Bogan, S. A., Antonarakis, A. S., & Moorcroft, P. R. (2019). Imaging spectrometryderived estimates of regional ecosystem composition for the Sierra Nevada, California. *Remote Sensing of Environment*, 228 (June 2017), 14–30. Retrieved from https://doi.org/10.1016/j.rse.2019.03.031 doi: 10.1016/j.rse.2019.03.031
- Braghiere, R. K., Quaife, T., Black, E., He, L., & Chen, J. M. (2019, 11). Underestimation of Global Photosynthesis in Earth System Models Due to Representation of Vegetation Structure. *Global Biogeochemical Cycles*, 2018GB006135. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1029/ 2018GB006135 doi: 10.1029/2018GB006135

21698961, ja, Downloaded from https://agupubs.onlinelibary.wiley.com/ubi/10.10292023/2007421 by Univ or California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://onlinelibary.wiley.com/terms-and-conditions) on Wiley Online Libary of rules of use; OA articles are govened by the applicable Creative Commo

- Brienen, R. J. W., Phillips, O. L., Feldpausch, T. R., Gloor, E., Baker, T. R., Lloyd, J., ... Zagt, R. J. (2015, 3). Long-term decline of the Amazon carbon sink. Nature, 519(7543), 344-348. Retrieved from http://www.nature.com/doifinder/10.1038/nature14283 doi: 10.1038/nature14283
- Cadotte, M. W., Carscadden, K., & Mirotchnick, N. (2011, 10). Beyond species: functional diversity and the maintenance of ecological processes and services. Journal of Applied Ecology, 48(5), 1079–1087. Retrieved from http://doi.wiley.com/10.1111/j.1365-2664.2011.02048.x doi: 10.1111/j.1365-2664.2011.02048.x
- Cardinale, B. J., Duffy, J. E., Gonzalez, A., Hooper, D. U., Perrings, C., Venail, P.,... Naeem, S. (2012, 6). Biodiversity loss and its impact on humanity. Nature, 486(7401), 59-67. Retrieved from http://www.nature.com/ articles/nature11373http://www.nature.com/articles/nature11148 doi: 10.1038/nature11148
- Cardinale, B. J., Ives, A. R., & Inchausti, P. (2004). Effects of species diversity on the primary productivity of ecosystems: Extending our spatial and temporal scales of inference. Oikos, 104(3), 437–450. doi: 10.1111/j.0030-1299.2004.13254.x
- Cardinale, B. J., Matulich, K. L., Hooper, D. U., Byrnes, J. E., Duffy, E., Gamfeldt, L., ... Gonzalez, A. (2011, 3). The functional role of producer diversity in ecosystems. American Journal of Botany, 98(3), 572–592. Retrieved from http://www.amjbot.org/cgi/doi/10.3732/ajb.1000364 doi: 10.3732/ajb.1000364
- Cawse-Nicholson, K., Townsend, P. A., Schimel, D., Assiri, A. M., Blake, P. L., Buongiorno, M. F., ... Zhang, Q. (2021, 5). NASA's surface biology

and geology designated observable: A perspective on surface imaging algorithms. *Remote Sensing of Environment*, 257, 112349. Retrieved from https://linkinghub.elsevier.com/retrieve/pii/S0034425721000675 doi: 10.1016/j.rse.2021.112349

- Caylor, K., Dowty, P., Shugart, H., & Ringrose, S. (2004, 3). Relationship between small-scale structural variability and simulated vegetation productivity across a regional moisture gradient in southern Africa. *Global Change Biology*, 10(3), 374–382. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1111/j.1365-2486.2003.00704.xhttps://onlinelibrary.wiley.com/doi/10.1111/j.1365-2486.2003.00704.x doi: 10.1111/j.1365-2486.2003.00704.x
- Ceballos, G., Ehrlich, P. R., Barnosky, A. D., García, A., Pringle, R. M., & Palmer, T. M. (2015). Accelerated modern human–induced species losses: Entering the sixth mass extinction. *Science Advances*, 1(5), e1400253. doi: 10.1126/sciadv.1400253
- Chamagne, J., Tanadini, M., Frank, D., Matula, R., Paine, C. E., Philipson, C. D., ... Hector, A. (2017). Forest diversity promotes individual tree growth in central European forest stands. *Journal of Applied Ecology*, 54(1), 71–79. doi: 10.1111/1365-2664.12783
- Chisholm, R. A., Muller-Landau, H. C., Abdul Rahman, K., Bebber, D. P., Bin, Y., Bohlman, S. A., ... Zimmerman, J. K. (2013, 9). Scale-dependent relationships between tree species richness and ecosystem function in forests. *Journal* of Ecology, 101(5), 1214–1224. Retrieved from http://doi.wiley.com/ 10.1111/1365-2745.12132 doi: 10.1111/1365-2745.12132
- Craven, D., Eisenhauer, N., Pearse, W. D., Hautier, Y., Isbell, F., Roscher, C., ... Manning, P. (2018). Multiple facets of biodiversity drive the diversity-stability relationship. Nature Ecology & Evolution. Retrieved from http://www.nature.com/articles/s41559-018-0647-7 doi: 10.1038/s41559-018-0647-7
- Czyz, E. A., Guillén Escribà, C., Wulf, H., Tedder, A., Schuman, M. C., Schneider, F. D., & Schaepman, M. E. (2020, 7). Intraspecific genetic variation of a Fagus sylvatica population in a temperate forest derived from airborne imaging spectroscopy time series. *Ecology and Evolution*, 10(14), 7419–7430. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1002/ece3.6469 doi: 10.1002/ece3.6469
- Czyż, E. A., Schmid, B., Hueni, A., Eppinga, M. B., Schuman, M. C., Schneider,
 F. D., ... Schaepman, M. E. (2023, 1). Genetic constraints on temporal variation of airborne reflectance spectra and their uncertainties over a temperate forest. *Remote Sensing of Environment*, 284 (February 2022), 113338. Retrieved from https://linkinghub.elsevier.com/retrieve/pii/S0034425722004448 doi: 10.1016/j.rse.2022.113338
- Damm, A., Paul-Limoges, E., Kükenbrink, D., Bachofen, C., & Morsdorf, F. (2020, 4). Remote sensing of forest gas exchange: Considerations derived from a tomographic perspective. *Global Change Biology*, 26(4), 2717-2727. Retrieved from https://onlinelibrary.wiley.com/doi/10.1111/gcb.15007 doi: 10.1111/gcb.15007
- Díaz, S., & Cabido, M. (2001). Vive la différence: plant functional diversity matters to ecosystem processes. Trends in Ecology & Evolution, 16(11), 646-655.
 Retrieved from http://www.sciencedirect.com/science/article/pii/
 S0169534701022832 doi: http://dx.doi.org/10.1016/S0169-5347(01)02283-2
- Díaz, S., Kattge, J., Cornelissen, J. H. C., Wright, I. J., Lavorel, S., Dray, S., ... Gorné, L. D. (2015). The global spectrum of plant form and function. *Nature*, 529(7585), 1–17. Retrieved from http://dx.doi.org/10.1038/nature16489 doi: 10.1038/nature16489
- Díaz, S., Settele, J., Brondízio, E. S., Ngo, H. T., Agard, J., Arneth, A., ... Zayas, C. N. (2019a). Pervasive human-driven decline of life on Earth

points to the need for transformative change. Science, 366(6471). doi: 10.1126/science.aax3100

- Díaz, S., Settele, J., Brondízio, E. S., Ngo, H. T., Agard, J., Arneth, A., ... Zayas, C. N. (2019b, 12). Pervasive human-driven decline of life on Earth points to the need for transformative change. *Science*, 366(6471), eaax3100. Retrieved from https://www.sciencemag.org/lookup/doi/10.1126/science.aax3100 doi: 10.1126/science.aax3100
- Dolezal, J., Fibich, P., Altman, J., Leps, J., Uemura, S., Takahashi, K., & Hara, T. (2020, 11). Determinants of ecosystem stability in a diverse temperate forest. *Oikos*, 129(11), 1692–1703. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1111/oik.05876https://onlinelibrary.wiley.com/doi/ 10.1111/oik.07379 doi: 10.1111/oik.07379
- Duffy, J. E., Godwin, C. M., & Cardinale, B. J. (2017, 9). Biodiversity effects in the wild are common and as strong as key drivers of productivity. Nature, 549(7671), 261–264. Retrieved from http://www.nature.com/doifinder/ 10.1038/nature23886 doi: 10.1038/nature23886
- Durán, S. M., Martin, R. E., Díaz, S., Maitner, B. S., Malhi, Y., Salinas, N., ... Enquist, B. J. (2019, 12). Informing trait-based ecology by assessing remotely sensed functional diversity across a broad tropical temperature gradient. Science Advances, 5(12), eaaw8114. Retrieved from http:// advances.sciencemag.org/lookup/doi/10.1126/sciadv.aaw8114https:// advances.sciencemag.org/lookup/doi/10.1126/sciadv.aaw8114 doi: 10.1126/sciadv.aaw8114
- Etzold, S., Buchmann, N., & Eugster, W. (2010, 8). Contribution of advection to the carbon budget measured by eddy covariance at a steep mountain slope forest in Switzerland. *Biogeosciences*, 7(8), 2461–2475. Retrieved from https://bg.copernicus.org/articles/7/2461/2010/ doi: 10.5194/bg-7-2461-2010
- Fawcett, D., Verhoef, W., Schläpfer, D., Schneider, F., Schaepman, M., & Damm,
 A. (2018, 1). Advancing retrievals of surface reflectance and vegetation indices over forest ecosystems by combining imaging spectroscopy, digital object models, and 3D canopy modelling. *Remote Sensing of Environment*, 204 (October 2017), 583-595. Retrieved from https://doi.org/10.1016/j.rse.2017.09
 .040https://linkinghub.elsevier.com/retrieve/pii/S0034425717304637
 doi: 10.1016/j.rse.2017.09.040
- Fischer, D. G., Wimp, G. M., Hersch-Green, E., Bangert, R. K., LeRoy, C. J., Bailey, J. K., ... Whitham, T. G. (2016). Tree Genetics Strongly Affect Forest Productivity, but Intraspecific Diversity-Productivity Relationships Do Not. *Functional Ecology*, 1–10. Retrieved from http://doi.wiley.com/10.1111/1365-2435.12733 doi: 10.1111/1365-2435.12733
- Fisher, R. A., & Koven, C. D. (2020). Perspectives on the Future of Land Surface Models and the Challenges of Representing Complex Terrestrial Systems. Journal of Advances in Modeling Earth Systems, 12(4). doi: 10.1029/2018MS001453
- Fisher, R. A., Koven, C. D., Anderegg, W. R. L., Christoffersen, B. O., Dietze,
 M. C., Farrior, C. E., ... Moorcroft, P. R. (2018, 1). Vegetation demographics in Earth System Models: A review of progress and priorities. *Global Change Biology*, 24(1), 35–54. Retrieved from http://doi.wiley.com/10.1111/ gcb.13910 doi: 10.1111/gcb.13910
- FOAG. (1996). Bodenkarte Baden, Landeskarte der Schweiz 1:25'000, Blatt 1070, Swiss Federal Office for Agriculture FOAG. Swiss Federal Office for Agriculture FOAG.
- Gonzalez, A., Germain, R. M., Srivastava, D. S., Filotas, E., Dee, L. E., Gravel, D.,
 ... Loreau, M. (2020). Scaling-up biodiversity-ecosystem functioning research. *Ecology Letters*, 23(4), 757–776. doi: 10.1111/ele.13456

- Grace, J. B., Anderson, T. M., Seabloom, E. W., Borer, E. T., Adler, P. B., Harpole, W. S., ... Smith, M. D. (2016, 1). Integrative modelling reveals mechanisms linking productivity and plant species richness. Nature, 529(7586), 390-393. Retrieved from http://dx.doi.org/10.1038/nature16524%
 5Cnpapers2://publication/doi/10.1038/nature16524http://www.nature.com/doifinder/10.1038/nature16524
- Guillén-Escribà, C., Schneider, F. D., Schmid, B., Tedder, A., Morsdorf, F., Furrer, R., ... Schaepman, M. E. (2021, 8). Remotely sensed between-individual functional trait variation in a temperate forest. *Ecology and Evolution*, 11(16), 10834–10867. Retrieved from https://onlinelibrary.wiley.com/doi/ 10.1002/ece3.7758 doi: 10.1002/ece3.7758
- Hagan, J. G., Vanschoenwinkel, B., & Gamfeldt, L. (2021, 12). We should not necessarily expect positive relationships between biodiversity and ecosystem functioning in observational field data. *Ecology Letters*, 24(12), 2537–2548. Retrieved from https://onlinelibrary.wiley.com/doi/10.1111/ele.13874 doi: 10.1111/ele.13874
- Hautier, Y., Isbell, F., Borer, E. T., Seabloom, E. W., Harpole, W. S., Lind, E. M.,
 ... Hector, A. (2018). Local loss and spatial homogenization of plant diversity reduce ecosystem multifunctionality. Nature Ecology and Evolution, 2(1), 50–56. Retrieved from http://dx.doi.org/10.1038/s41559-017-0395-0
 doi: 10.1038/s41559-017-0395-0
- Heiri, C., Wolf, A., Rohrer, L., & Bugmann, H. (2009). Forty years of natural dynamics in Swiss beech forests: Structure, composition, and the influence of former management. *Ecological Applications*, 19(7), 1920–1934. doi: 10.1890/08-0516.1
- Helfenstein, I. S., Schneider, F. D., Schaepman, M. E., & Morsdorf, F. (2022, 6). Assessing biodiversity from space: Impact of spatial and spectral resolution on trait-based functional diversity. *Remote Sensing of Environment*, 275 (March), 113024. Retrieved from https://doi.org/10.1016/j.rse.2022.113024https://linkinghub.elsevier.com/retrieve/pii/S0034425722001389 doi: 10.1016/j.rse.2022.113024
- Hong, P., Schmid, B., De Laender, F., Eisenhauer, N., Zhang, X., Chen, H., ... Wang, S. (2022, 2). Biodiversity promotes ecosystem functioning despite environmental change. *Ecology Letters*, 25(2), 555–569. Retrieved from https://onlinelibrary.wiley.com/doi/10.1111/ele.13936 doi: 10.1111/ele.13936
- Hooper, D. U., Chapin, F. S., Ewel, J. J., Hector, A., Inchausti, P., Lavorel, S., ... Wardle, D. A. (2005, 2). Effects of biodiversity on ecosystem functioning: a consensus of current knowledge. *Ecological Monographs*, 75(1), 3-35. Retrieved from http://www.esajournals.org/doi/pdf/10.1890/04-0922http:// doi.wiley.com/10.1890/04-0922http://dx.doi.org/10.1890/04-0922 doi: 10.1890/04-0922
- Huang, Y., Chen, Y., Castro-Izaguirre, N., Baruffol, M., Brezzi, M., Lang, A., ... Schmid, B. (2018). Impacts of species richness on productivity in a large-scale subtropical forest experiment. *Science*, 362(6410), 80-83. Retrieved from http://www.sciencemag.org/lookup/doi/10.1126/science.aat6405 doi: 10.1126/science.aat6405
- Hubau, W., Lewis, S. L., Phillips, O. L., Affum-Baffoe, K., Beeckman, H., Cuní-Sanchez, A., ... Zemagho, L. (2020, 3). Asynchronous carbon sink saturation in African and Amazonian tropical forests. *Nature*, 579(7797), 80–87. Retrieved from http://www.nature.com/articles/s41586-020-2035-0 doi: 10.1038/s41586-020-2035-0
- Hueni, A., Biesemans, J., Meuleman, K., Dell'Endice, F., Schlapfer, D., Odermatt,
 D., ... Itten, K. I. (2009, 1). Structure, Components, and Interfaces of the Airborne Prism Experiment (APEX) Processing and Archiving Facility.

IEEE Transactions on Geoscience and Remote Sensing, 47(1), 29–43. doi: 10.1109/TGRS.2008.2005828

- Hueni, A., Lenhard, K., Baumgartner, A., & Schaepman, M. E. (2013, 11). Airborne Prism Experiment Calibration Information System. *IEEE Transactions on Geoscience and Remote Sensing*, 51(11), 5169–5180. doi: 10.1109/TGRS .2013.2246575
- Hurtt, G. C., Fisk, J., Thomas, R. Q., Dubayah, R., Moorcroft, P. R., & Shugart,
 H. H. (2010). Linking models and data on vegetation structure. Journal of Geophysical Research, 115, 1–11. doi: 10.1029/2009JG000937
- Isbell, F., Craven, D., Connolly, J., Loreau, M., Schmid, B., Beierkuhnlein, C., ... Eisenhauer, N. (2015, 10). Biodiversity increases the resistance of ecosystem productivity to climate extremes. Nature, 526(7574), 574–577. Retrieved from http://www.nature.com/doifinder/10.1038/nature15374 doi: 10.1038/nature15374
- Isbell, F., Gonzalez, A., Loreau, M., Cowles, J., Díaz, S., Hector, A., ... Larigauderie, A. (2017, 5). Linking the influence and dependence of people on biodiversity across scales. Nature, 546(7656), 65–72. Retrieved from http://www.nature.com/doifinder/10.1038/nature22899 doi: 10.1038/nature22899
- Ishii, H. T., Tanabe, S., & Hiura, T. (2004). Exploring the relationships among canopy structure, stand productivity, and biodiversity of temperature forest ecosystems. *Forest Science*, 50(3), 342–355.
- Jin, W., He, H. S., Thompson, F. R., Wang, W. J., Fraser, J. S., Shifley, S. R., ... Dijak, W. D. (2017, 3). Future forest aboveground carbon dynamics in the central United States: the importance of forest demographic processes. *Scientific Reports*, 7(1), 41821. Retrieved from http://www.nature.com/articles/ srep41821 doi: 10.1038/srep41821
- Jochum, M., Fischer, M., Isbell, F., Roscher, C., van der Plas, F., Boch, S., ... Manning, P. (2020, 8). The results of biodiversity-ecosystem functioning experiments are realistic. Nature Ecology & Evolution. Retrieved from http://www.nature.com/articles/s41559-020-1280-9 doi: 10.1038/s41559-020-1280-9
- Kaartinen, H., Hyyppä, J., Yu, X., Vastaranta, M., Hyyppä, H., Kukko, A., ... Wu, J.-C. C. (2012). An international comparison of individual tree detection and extraction using airborne laser scanning. *Remote Sensing*, 4(4), 950–974. doi: 10.3390/rs4040950
- Kljun, N., Calanca, P., Rotach, M. W., & Schmid, H. P. (2015). A simple twodimensional parameterisation for Flux Footprint Prediction (FFP). Geoscientific Model Development, 8(11), 3695–3713. doi: 10.5194/gmd-8-3695-2015
- Kükenbrink, D., Schneider, F. D., Leiterer, R., Schaepman, M. E., & Morsdorf,
 F. (2017, 6). Quantification of hidden canopy volume of airborne laser scanning data using a voxel traversal algorithm. Remote Sensing of Environment, 194 (Special Issue of the SilviLaser 2015 Conference), 424–436. Retrieved from http://linkinghub.elsevier.com/retrieve/pii/S0034425716303959https://linkinghub.elsevier.com/retrieve/pii/S0034425716303959 doi: 10.1016/j.rse.2016.10.023
- Kükenbrink, D., Schneider, F. D., Schmid, B., Gastellu-Etchegorry, J.-P., Schaepman, M. E., & Morsdorf, F. (2021, 1). Modelling of three-dimensional, diurnal light extinction in two contrasting forests. *Agricultural and Forest Meteorology*, 296 (November 2020), 108230. Retrieved from https://linkinghub.elsevier.com/retrieve/pii/S0168192320303324https://doi.org/10.1016/j.agrformet.2020.108230 doi: 10.1016/j.agrformet.2020.108230

Levine, N. M., Zhang, K., Longo, M., Baccini, A., Phillips, O. L., Lewis, S. L., ... Moorcroft, P. R. (2016, 1). Ecosystem heterogeneity determines the ecological resilience of the Amazon to climate change. *Proceedings* of the National Academy of Sciences, 113(3), 793-797. Retrieved from http://www.pnas.org/lookup/doi/10.1073/pnas.1511344112 doi: 10.1073/pnas.1511344112

- Liang, J., Crowther, T. W., Picard, N., Wiser, S., Zhou, M., Alberti, G., ... Reich, P. B. (2016, 10). Positive biodiversity-productivity relationship predominant in global forests. *Science*, 354 (6309), aaf8957-aaf8957. Retrieved from http://www.sciencemag.org/cgi/doi/10.1126/science.aaf8957 doi: 10.1126/science.aaf8957
- Liu, J., Skidmore, A. K., Jones, S., Wang, T., Heurich, M., Zhu, X., & Shi, Y.
 (2018). Large off-nadir scan angle of airborne LiDAR can severely affect the estimates of forest structure metrics. *ISPRS Journal of Photogrammetry and Remote Sensing*, 136(February), 13–25. Retrieved from https://doi.org/10.1016/j.isprsjprs.2017.12.004 doi: 10.1016/j.isprsjprs.2017.12.004
- Liu, X., Trogisch, S., He, J.-S., Niklaus, P. A., Bruelheide, H., Tang, Z., ... Ma, K. (2018, 8). Tree species richness increases ecosystem carbon storage in subtropical forests. *Proceedings of the Royal Society B: Biological Sciences*, 285(1885), 20181240. Retrieved from https://royalsocietypublishing.org/doi/ 10.1098/rspb.2018.1240 doi: 10.1098/rspb.2018.1240
- Longo, M., Knox, R. G., Levine, N. M., Alves, L. F., Bonal, D., Camargo, P. B., ... Moorcroft, P. R. (2018, 8). Ecosystem heterogeneity and diversity mitigate Amazon forest resilience to frequent extreme droughts. New Phytologist, 219(3), 914–931. Retrieved from http://doi.wiley.com/10.1111/nph.15185 doi: 10.1111/nph.15185
- Longo, M., Knox, R. G., Levine, N. M., Swann, A. L. S., Medvigy, D. M., Dietze, M. C., ... Moorcroft, P. R. (2019, 10). The biophysics, ecology, and biogeochemistry of functionally diverse, vertically and horizontally heterogeneous ecosystems: the Ecosystem Demography model, version 2.2 – Part 2: Model evaluation for tropical South America. Geoscientific Model Development, 12(10), 4347–4374. Retrieved from https://gmd.copernicus.org/articles/ 12/4347/2019/https://gmd.copernicus.org/articles/12/4309/2019/ doi: 10.5194/gmd-12-4347-2019
- Longo, M., Knox, R. G., Medvigy, D. M., Levine, N. M., Dietze, M. C., Kim, Y., ... Moorcroft, P. R. (2019, 10). The biophysics, ecology, and biogeochemistry of functionally diverse, vertically and horizontally heterogeneous ecosystems: the Ecosystem Demography model, version 2.2 – Part 1: Model description. *Geoscientific Model Development*, 12(10), 4309–4346. Retrieved from https://gmd.copernicus.org/articles/12/4309/2019/ doi: 10.5194/gmd-12-4309-2019
- Longo, M., Saatchi, S., Keller, M., Bowman, K., Ferraz, A., Moorcroft, P. R., ... Vincent, G. (2020, 8). Impacts of Degradation on Water, Energy, and Carbon Cycling of the Amazon Tropical Forests. Journal of Geophysical Research: Biogeosciences, 125(8), 1–27. Retrieved from https://onlinelibrary.wiley.com/doi/10.1029/2020JG005677 doi: 10.1029/2020JG005677
- Loreau, M. (2000, 10). Biodiversity and ecosystem functioning: recent theoretical advances. Oikos, 91(1), 3-17. Retrieved from http://doi.wiley.com/10 .1034/j.1600-0706.2000.910101.x doi: 10.1034/j.1600-0706.2000.910101.x
- Loreau, M., & de Mazancourt, C. (2013). Biodiversity and ecosystem stability: A synthesis of underlying mechanisms. *Ecology Letters*, 16(SUPPL.1), 106–115. doi: 10.1111/ele.12073
- Loreau, M., Naeem, S., Inchausti, P., Bengtsson, J., Grime, J. P., Hector, A., ... Wardle, D. A. (2001). Biodiversity and Ecosystem Functioning: Current Knowledge and Future Challenges. *Science*, 294(5543), 804–808. Retrieved from http://www.sciencemag.org/content/294/5543/804.abstract doi: 10.1126/science.1064088

- Mason, N. W. H., & de Bello, F. (2013, 9). Functional diversity: a tool for answering challenging ecological questions. Journal of Vegetation Science, 24(5), 777-780. Retrieved from http://dx.doi.org/10.1111/jvs.12097http://doi .wiley.com/10.1111/jvs.12097 doi: 10.1111/jvs.12097
- Medvigy, D., Wang, G., Zhu, Q., Riley, W. J., Trierweiler, A. M., Waring, B.,
 ... Powers, J. S. (2019). Observed variation in soil properties can drive large variation in modelled forest functioning and composition during tropical forest secondary succession. New Phytologist, 223(4), 1820–1833. doi: 10.1111/nph.15848
- Medvigy, D., Wofsy, S. C., Munger, J. W., Hollinger, D. Y., & Moorcroft, P. R. (2009, 1). Mechanistic scaling of ecosystem function and dynamics in space and time: Ecosystem Demography model version 2. Journal of Geophysical Research, 114 (G1), G01002. Retrieved from http://doi.wiley.com/10.1029/2008JG000812 doi: 10.1029/2008JG000812
- Medvigy, D., Wofsy, S. C., Munger, J. W., & Moorcroft, P. R. (2010, 5). Responses of terrestrial ecosystems and carbon budgets to current and future environmental variability. *Proceedings of the National Academy of Sciences*, 107(18), 8275–8280. Retrieved from http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2889511&tool=pmcentrez&rendertype=abstracthttp://www.pnas.org/cgi/doi/10.1073/pnas.0912032107 doi: 10.1073/pnas.0912032107
- Meunier, F., Krishna Moorthy, S. M., Peaucelle, M., Calders, K., Terryn, L., Verbruggen, W., ... Verbeeck, H. (2022, 6). Using terrestrial laser scanning to constrain forest ecosystem structure and functions in the Ecosystem Demography model (ED2.2). Geoscientific Model Development, 15(12), 4783–4803. Retrieved from https://gmd.copernicus.org/articles/15/4783/2022/doi: 10.5194/gmd-15-4783-2022
- Meunier, F., Verbeeck, H., Cowdery, B., Schnitzer, S. A., Smith-Martin, C. M., Powers, J. S., ... Dietze, M. (2021, 1). Unraveling the relative role of light and water competition between lianas and trees in tropical forests: A vegetation model analysis. *Journal of Ecology*, 109(1), 519–540. Retrieved from https://onlinelibrary.wiley.com/doi/10.1111/1365-2745.13540 doi: 10.1111/1365-2745.13540

21698961, ja, Downloaded from https://agupubs.onlinelibary.wiley.com/ubi/10.10292023/2007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lawrence Berkeley National Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292023/0007421 by Univ of California Lab, Wiley Online Libary on (1905/2023). See the Terms and Conditions (https://allinelibary.wiley.com/ubi/10.10292

- Mina, M., Huber, M. O., Forrester, D. I., Thürig, E., & Rohner, B. (2017, 9). Multiple factors modulate tree growth complementarity in Central European mixed forests. *Journal of Ecology*(May), 1–14. Retrieved from http:// doi.wiley.com/10.1111/1365-2745.12846 doi: 10.1111/1365-2745.12846
- Moorcroft, P. R., Hurtt, G. C., & Pacala, S. W. (2001). A method for scaling vegetation dynamics: The ecosystem demography model (ED). *Ecological Mono*graphs, 71(4), 557–586. doi: 10.1890/0012-9615(2001)071
- Mori, A. S. (2020, 9). Advancing nature-based approaches to address the biodiversity and climate emergency. *Ecology Letters*, ele.13594. Retrieved from https://onlinelibrary.wiley.com/doi/10.1111/ele.13594 doi: 10.1111/ ele.13594
- Morin, X., Fahse, L., Scherer-Lorenzen, M., & Bugmann, H. (2011, 12). Tree species richness promotes productivity in temperate forests through strong complementarity between species. *Ecology Letters*, 14(12), 1211–1219. Retrieved from http://doi.wiley.com/10.1111/j.1461-0248.2011.01691.x doi: 10.1111/j.1461-0248.2011.01691.x
- Morsdorf, F., Meier, E., Kötz, B., Itten, K. I., Dobbertin, M., & Allgöwer, B. (2004). LIDAR-based geometric reconstruction of boreal type forest stands at single tree level for forest and wildland fire management. *Remote Sensing of Environment*, 92(3), 353–362.
- Morsdorf, F., Schneider, F. D., Gullien, C., Kükenbrink, D., Leiterer, R., & Schaepman, M. E. (2020). The Laegeren Site: An Augmented Forest Laboratory.

This article is protected by copyright. All rights reserved.

In J. Cavender-Bares, J. A. Gamon, & P. A. Townsend (Eds.), Remote sensing of plant biodiversity (pp. 83–104). Cham: Springer International Publishing. Retrieved from https://doi.org/10.1007/978-3-030-33157-3_4http://link.springer.com/10.1007/978-3-030-33157-3_4 doi: $10.1007/978-3-030-33157-3\{\end{textbackslash}\{\]4$

- Morton, D. C., Rubio, J., Cook, B. D., Gastellu-Etchegorry, J.-P., Longo, M., Choi, H., ... Keller, M. (2016, 4). Amazon forest structure generates diurnal and seasonal variability in light utilization. *Biogeosciences*, 13(7), 2195–2206. Retrieved from http://www.biogeosciences.net/13/2195/2016/ doi: 10.5194/bg-13-2195-2016
- Oehri, J., Bürgin, M., Schmid, B., & Niklaus, P. A. (2020, 5). Local and landscapelevel diversity effects on forest functioning. *PLOS ONE*, 15(5), e0233104. Retrieved from https://dx.plos.org/10.1371/journal.pone.0233104 doi: 10 .1371/journal.pone.0233104
- Paquette, A., & Messier, C. (2011, 1). The effect of biodiversity on tree productivity: from temperate to boreal forests. *Global Ecology and Biogeography*, 20(1), 170–180. Retrieved from http://doi.wiley.com/10.1111/j.1466-8238.2010 .00592.xhttp://dx.doi.org/10.1111/j.1466-8238.2010.00592.x doi: 10 .1111/j.1466-8238.2010.00592.x
- Paul-Limoges, E., Damm, A., Hueni, A., Liebisch, F., Eugster, W., Schaepman, M. E., & Buchmann, N. (2018, 12). Effect of environmental conditions on sun-induced fluorescence in a mixed forest and a cropland. *Remote Sensing* of Environment, 219 (November 2017), 310-323. Retrieved from https:// doi.org/10.1016/j.rse.2018.10.018https://linkinghub.elsevier.com/ retrieve/pii/S0034425718304735 doi: 10.1016/j.rse.2018.10.018
- Paul-Limoges, E., Wolf, S., Eugster, W., Hörtnagl, L., & Buchmann, N. (2017). Below-canopy contributions to ecosystem CO2fluxes in a temperate mixed forest in Switzerland. Agricultural and Forest Meteorology, 247(August), 582–596. Retrieved from http://dx.doi.org/10.1016/j.agrformet.2017.08.011 doi: 10.1016/j.agrformet.2017.08.011
- Paul-Limoges, E., Wolf, S., Schneider, F. D., Longo, M., Moorcroft, P., Gharun, M., & Damm, A. (2020, 1). Partitioning evapotranspiration with concurrent eddy covariance measurements in a mixed forest. Agricultural and Forest Meteorology, 280(October 2019), 107786. Retrieved from https://doi.org/10.1016/ j.agrformet.2019.107786https://linkinghub.elsevier.com/retrieve/ pii/S0168192319304022 doi: 10.1016/j.agrformet.2019.107786
- Petchey, O. L., Hector, A., & Gaston, K. J. (2004, 3). HOW DO DIFFERENT MEASURES OF FUNCTIONAL DIVERSITY PERFORM? *Ecology*, 85(3), 847-857. Retrieved from https://esajournals.onlinelibrary.wiley.com/ doi/abs/10.1890/03-0226http://doi.wiley.com/10.1890/03-0226 doi: 10 .1890/03-0226
- Peters, R. (1992). Ecology of beech forests in the northern hemisphere (Doctoral dissertation, Wageningen). Retrieved from https://edepot.wur.nl/200953
- Poorter, L., van der Sande, M. T., Thompson, J., Arets, E. J. M. M., Alarcón, A., Álvarez-Sánchez, J., ... Peña-Claros, M. (2015, 11). Diversity enhances carbon storage in tropical forests. Global Ecology and Biogeography, 24(11), 1314– 1328. Retrieved from https://onlinelibrary.wiley.com/doi/10.1111/ geb.12364 doi: 10.1111/geb.12364
- Ratcliffe, S., Wirth, C., Jucker, T., van der Plas, F., Scherer-Lorenzen, M., Verheyen, K., ... Baeten, L. (2017, 11). Biodiversity and ecosystem functioning relations in European forests depend on environmental context. *Ecology Letters*, 20(11), 1414–1426. Retrieved from http://doi.wiley.com/10.1111/ ele.12849 doi: 10.1111/ele.12849
- Reich, P. B., Wright, I. J., Cavender-Bares, J., Craine, J. M., Oleksyn, J., Westoby,M., & Walters, M. B. (2003). The Evolution of Plant Functional Variation:

Traits, Spectra, and Strategies. International Journal of Plant Sciences, 164(S3), S143-S164. Retrieved from https://doi.org/10.1086/374368 doi: 10.1086/374368

- Rödig, E., Cuntz, M., Rammig, A., Fischer, R., Taubert, F., & Huth, A. (2018, 5). The importance of forest structure for carbon fluxes of the Amazon rainforest. *Environmental Research Letters*, 13(5), 054013. Retrieved from https://iopscience.iop.org/article/10.1088/1748-9326/aabc61 doi: 10.1088/1748-9326/aabc61
- Roscher, C., Gubsch, M., Lipowsky, A., Schumacher, J., Weigelt, A., Buchmann, N., ... Schmid, B. (2018, 6). Trait means, trait plasticity and trait differences to other species jointly explain species performances in grasslands of varying diversity. *Oikos*, 127(6), 865–865. Retrieved from http://doi.wiley.com/10.1111/oik.04815 doi: 10.1111/oik.04815
- Rosenfeld, J. S. (2002, 7). Functional redundancy in ecology and conservation. *Oikos*, 98(1), 156-162. Retrieved from http://doi.wiley.com/10.1034/ j.1600-0706.2002.980116.x doi: 10.1034/j.1600-0706.2002.980116.x
- Sabatini, F. M., de Andrade, R. B., Paillet, Y., Ódor, P., Bouget, C., Campagnaro, T., ... Burrascano, S. (2019, 2). Trade-offs between carbon stocks and biodiversity in European temperate forests. *Global Change Biology*, 25(2), 536-548. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1111/ gcb.14503 doi: 10.1111/gcb.14503
- Sakschewski, B., von Bloh, W., Boit, A., Poorter, L., Pena-Claros, M., Heinke, J., ... Thonicke, K. (2016). Resilience of Amazon forests emerges from plant trait diversity. *Nature Clim. Change*, 6(11), 1032–1036. Retrieved from http://dx.doi.org/10.1038/nclimate3109
- Sakschewski, B., von Bloh, W., Boit, A., Rammig, A., Kattge, J., Poorter, L., ... Thonicke, K. (2015). Leaf and stem economics spectra drive diversity of functional plant traits in a dynamic global vegetation model. *Global Change Biology*, 21(7), 2711–2725. doi: 10.1111/gcb.12870
- Schaepman, M. E., Jehle, M., Hueni, A., D'Odorico, P., Damm, A., Weyermann, J., ... Itten, K. I. (2015, 3). Advanced radiometry measurements and Earth science applications with the Airborne Prism Experiment (APEX). *Remote Sensing of Environment*, 158(1), 207-219. Retrieved from http://dx.doi.org/ 10.1016/j.rse.2014.11.014http://linkinghub.elsevier.com/retrieve/ pii/S0034425714004568https://linkinghub.elsevier.com/retrieve/pii/ S0034425714004568 doi: 10.1016/j.rse.2014.11.014
- Schimel, D., Schneider, F. D., & JPL Carbon and Ecosystem Participants. (2019, 10). Flux towers in the sky: global ecology from space. New Phytologist, 224(2), 570-584. Retrieved from https://onlinelibrary.wiley.com/doi/ abs/10.1111/nph.15934 doi: 10.1111/nph.15934
- Schimel, D., Stephens, B. B., & Fisher, J. B. (2015, 1). Effect of increasing CO 2 on the terrestrial carbon cycle. Proceedings of the National Academy of Sciences, 112(2), 436-441. Retrieved from http://www.pnas.org/lookup/doi/ 10.1073/pnas.1407302112 doi: 10.1073/pnas.1407302112
- Schläpfer, D., & Richter, R. (2002). Geo-atmospheric processing of airborne imaging spectrometry data. Part 1: Parametric orthorectification. International Journal of Remote Sensing, 23(13), 2609–2630. Retrieved from http://dx.doi.org/ 10.1080/01431160110115825 doi: 10.1080/01431160110115825
- Schneider, F. D., Ferraz, A., Hancock, S., Duncanson, L. I., Dubayah, R. O., Pavlick, R. P., & Schimel, D. S. (2020, 10). Towards mapping the diversity of canopy structure from space with GEDI. Environmental Research Letters, 15(11), 115006. Retrieved from https://iopscience.iop.org/article/ 10.1088/1748-9326/ab9e99 doi: 10.1088/1748-9326/ab9e99
- Schneider, F. D., Kükenbrink, D., Schaepman, M. E., Schimel, D. S., & Morsdorf, F. (2019). Quantifying 3D structure and occlusion in dense tropical and tem-

perate forests using close-range LiDAR. Agricultural and Forest Meteorology, 268 (December 2018), 249–257. Retrieved from https://doi.org/10.1016/j.agrformet.2019.01.033 doi: 10.1016/j.agrformet.2019.01.033

- Schneider, F. D., Leiterer, R., Morsdorf, F., Gastellu-Etchegorry, J.-P., Lauret, N., Pfeifer, N., & Schaepman, M. E. (2014, 9). Simulating imaging spectrometer data: 3D forest modeling based on LiDAR and in situ data. Remote Sensing of Environment, 152, 235-250. Retrieved from https://linkinghub.elsevier.com/retrieve/pii/S0034425714002284 doi: 10.1016/j.rse.2014.06.015
- Schneider, F. D., Longo, M., Paul-Limoges, E., Scholl, V. M., Schmid, B., Morsdorf, F., ... Moorcroft, P. R. (2023). Data by Schneider et al. (2023) Remote sensing-based forest modeling reveals positive effects of functional diversity on productivity at local spatial scale. Retrieved from https://figshare.com/ articles/dataset/Data_by_Schneider_et_al_2023_Remote_sensing -based_forest_modeling_reveals_positive_effects_of_functional _diversity_on_productivity_at_local_spatial_scale/22705024 doi: 10.6084/m9.figshare.22705024.v1
- Schneider, F. D., Morsdorf, F., Schmid, B., Petchey, O. L., Hueni, A., Schimel, D. S., & Schaepman, M. E. (2017, 12). Mapping functional diversity from remotely sensed morphological and physiological forest traits. *Nature Communications*, 8(1), 1441. Retrieved from http://dx.doi.org/10.1038/s41467-017-01530-3 doi: 10.1038/s41467-017-01530-3
- Sheil, D., & Bongers, F. (2020). Interpreting forest diversity-productivity relationships: volume values, disturbance histories and alternative inferences. Forest Ecosystems, 7(1). doi: 10.1186/s40663-020-0215-x
- Silva Pedro, M., Rammer, W., & Seidl, R. (2016). A disturbance-induced increase in tree species diversity facilitates forest productivity. Landscape Ecology, 31(5), 989–1004. doi: 10.1007/s10980-015-0317-y
- Stöckli, R., Rutishauser, T., Baker, I., Liniger, M. A., & Denning, A. S. (2011, 8). A global reanalysis of vegetation phenology. *Journal of Geophysical Research*, 116(G3), G03020. Retrieved from http://doi.wiley.com/10.1029/2010JG001545 doi: 10.1029/2010JG001545
- Swenson, N. G. (2012). The Functional Ecology and Diversity of Tropical Tree Assemblages through Space and Time: From Local to Regional and from Traits to Transcriptomes. *ISRN Forestry*, 2012, 1–16. doi: 10.5402/2012/743617
- Tans, P., & Keeling, R. (2020). Trends in Atmospheric Carbon Dioxide, Mauna Loa, Hawaii. NOAA/ESRL (www.esrl.noaa.gov/gmd/ccgg/trends/); Scripps Institution of Oceanography (scrippsco2.ucsd.edu/).
- Thonicke, K., Billing, M., Bloh, W., Sakschewski, B., Niinemets, I., Peñuelas, J., ... Walz, A. (2020, 5). Simulating functional diversity of European natural forests along climatic gradients. *Journal of Biogeography*, 47(5), 1069–1085. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1111/jbi.13809 doi: 10.1111/jbi.13809
- Tilman, D., Isbell, F., & Cowles, J. M. (2014, 11). Biodiversity and Ecosystem Functioning. Annual Review of Ecology, Evolution, and Systematics, 45(1), 471–493. Retrieved from http://www.annualreviews.org/doi/10.1146/annurev-ecolsys-120213-091917 doi: 10.1146/annurev-ecolsys-120213-091917
- Tilman, D., Knops, J., Wedin, D., Reich, P., Ritchie, M., & Siemann, E. (1997, 8). The Influence of Functional Diversity and Composition on Ecosystem Processes. Science, 277(5330), 1300–1302. Retrieved from https:// www.science.org/doi/10.1126/science.277.5330.1300 doi: 10.1126/ science.277.5330.1300
- Tilman, D., Lehman, C. L., & Thomson, K. T. (1997, 3). Plant diversity and ecosystem productivity: Theoretical considerations. Proceedings of

the National Academy of Sciences, 94(5), 1857–1861. Retrieved from http://www.pnas.org/cgi/doi/10.1073/pnas.94.5.1857 doi: 10.1073/ pnas.94.5.1857

- Trisos, C. H., Merow, C., & Pigot, A. L. (2020, 4). The projected timing of abrupt ecological disruption from climate change. Nature, 580(7804), 496-501. Retrieved from http://dx.doi.org/10.1038/s41586-020 -2189-9http://www.nature.com/articles/s41586-020-2189-9 doi: 10.1038/s41586-020-2189-9
- van der Plas, F. (2019, 2). Biodiversity and ecosystem functioning in naturally assembled communities. Biological Reviews, 94(4), brv.12499. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1111/brv.12499 doi: 10 .1111/brv.12499
- Vickers, D., & Mahrt, L. (1997). Quality Control and Flux Sampling Problems for Tower and Aircraft Data. Journal of Atmospheric and Oceanic Technology, 14(3), 512–526. Retrieved from https://doi.org/10.1175/1520-0426(1997) 014%3C0512:QCAFSP%3E2.0.CO;2 doi: 10.1175/1520-0426(1997)014<0512: QCAFSP>2.0.CO;2
- Wang, X., Wiegand, T., Swenson, N. G., Wolf, A. T., Howe, R. W., Hao, Z., ... Yuan, Z. (2015). Mechanisms underlying local functional and phylogenetic beta diversity in two temperate forests. *Ecology*, 96(4), 1062–1073. doi: 10.1890/14-0392.1.sm
- Wang, Y., Hyyppa, J., Liang, X., Kaartinen, H., Yu, X., Lindberg, E., ... Alho, P. (2016). International Benchmarking of the Individual Tree Detection Methods for Modeling 3-D Canopy Structure for Silviculture and Forest Ecology Using Airborne Laser Scanning. *IEEE Transactions on Geoscience and Remote Sensing*, 54 (9), 5011–5027. doi: 10.1109/TGRS.2016.2543225
- Wang, Z., Chlus, A., Geygan, R., Ye, Z., Zheng, T., Singh, A., ... Townsend, P. A. (2020, 6). Foliar functional traits from imaging spectroscopy across biomes in eastern North America. New Phytologist, nph.16711. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1111/nph.16711 doi: 10.1111/nph.16711

21698961, ja, Downloaded from https://agupubs.onlinelibarary.wikey.com/ubi/10.10292023/G007421 by Univi or California Lawrence Berkeley National Lab, Wiley Online Library on [9095/2023]. See the Terms and Conditions (https://alinelibary.wikey.com/ubi/10.10292023) G007421 by Univi or California Lawrence Berkeley National Lab, Wiley Online Library on [9095/2023]. See the Terms and Conditions (https://alinelibary.wikey.com/ubi/10.10292023) G007421 by Univi or California Lawrence Berkeley National Lab, Wiley Online Library of the probability o

- Williams, L. J., Paquette, A., Cavender-Bares, J., Messier, C., & Reich, P. B. (2017, 3). Spatial complementarity in tree crowns explains overyielding in species mixtures. Nature Ecology & Evolution, 1(4), 0063. Retrieved from http://www.nature.com/articles/s41559-016-0063 doi: 10.1038/s41559-016-0063
- Xu, H., Cao, Y., Yu, D., Cao, M., He, Y., Gill, M., & Pereira, H. M. (2021). Ensuring effective implementation of the post-2020 global biodiversity targets. Nature Ecology and Evolution(iDiv). doi: 10.1038/s41559-020-01375-y
- Yachi, S., & Loreau, M. (1999, 2). Biodiversity and ecosystem productivity in a fluctuating environment: The insurance hypothesis. Proceedings of the National Academy of Sciences, 96(4), 1463–1468. Retrieved from http://www.pnas.org/cgi/doi/10.1073/pnas.96.4.1463 doi: 10.1073/pnas.96.4.1463